# Getting REDDy Understanding and Improving Domestic Policy Impacts on Forest Loss

Alexander Pfaff<sup>\*</sup> Gregory S. Amacher<sup>†</sup> Erin O. Sills<sup>‡</sup>

Working Paper EE 11-12 December 2011

\* Sanford School of Public Policy, Duke University

<sup>†</sup> Virginia Polytechnic Institute and State University

<sup>\*</sup> North Carolina State University

The Duke Environmental Economics Working Paper Series provides a forum for Duke faculty working in environmental and resource economics to disseminate their research. These working papers have not undergone peer review at the time of posting.





# **Getting REDDy:**

# understanding and improving domestic policy impacts on forest loss

Alexander Pfaff (Duke), Gregory S. Amacher (VA Tech) and Erin O. Sills (NC State)

September 25, 2011\*\*

### Abstract (185 words)

Many constraints upon REDD+ policies' ability to reduce forest loss are common across settings, inherent in the fact that agents making key choices respond also to other factors that influence the overall incentive to clear or to degrade a forest instead of conserving it. The record is mixed, at best, with regard to past public interventions to reduce forest loss, signaling the need to disseminate and to improve conceptual models of policy responses. We summarize 3 distinct models employed by economists to assess policy effectiveness: (1) producer profit maximization in choosing spatial extent and distribution of land uses, given complete markets; (2) rural household optimization given both incomplete markets and varied household assets and tastes; and (3) public optimization within interconnected choices about concessions, corruption and decentralization, all important for degradation ('D+' in REDD+). Each model's perspective on impact leads to a review of the evidence. We consider the impacts of forest-conservation and forest-relevant-development policies for the settings and decisions, and at the scales, for which each of the models best applies. Theory and evidence suggest options to increase the impacts of domestic REDD policies.

Keywords deforestation, forest degradation, parks, evaluation, REDD

JEL ### ### ### ###

<sup>&</sup>lt;sup>\*\*</sup>We thank Brian Murray and Lydia Olander of Duke's Nicholas Institute for initiating and for commenting on our first efforts to collect the empirical evidence that led to this paper, as well as Dan Zarin and the Packard Foundation for helpful comments on and financial support of those efforts plus our coauthors on the Packard/Nicholas Institute white paper citing a great deal of the empirical evidence that we refer to in this paper, i.e. Michael Coren, Kathleen Lawlor and Charlotte Streck. We also thank Suzi Kerr and the rest of our Symposium fellow authors for feedback on these ideas as well as participants at seminars in Copenhagen, for U.S. AID (Nicholas Institute), and in Costa Rica for U.S. policy makers (organized by OTS). All the errors that remain, despite all of the feedback, are fully our own.

### 1. Introduction

Policies to reduce tropical forest loss have had some impacts yet generally failed to substantially slow losses. The recognition that deforestation and forest degradation generate about one sixth of greenhouse-gas emissions has joined concern about losses of other eco-services, e.g. habitat and water quality, to motivate new global policy like REDD+ payments to reduce tropical forest loss. Domestic initiatives by countries that receive such global incentive payments should be informed by the mixed record of public interventions in addressing the local incentives in two key sectors.

Agriculture is the first key sector. Its expansion, including the supporting infrastructure, is the primary driver of tropical deforestation. Cleared forest land is almost always converted to crops or pasture driven by expected benefits of producing staple foods for local and national use plus internationally valued commodities like biofuels, pulp, and fiber. Deforestation pressure is exacerbated by government supports for agriculture from road investment to provision of cheap credit, sanitary programs, and easier access to titling for lands that have been cleared of forest.

Forest extraction is the second key sector, including logging and collection of wood fuel. Forest loss is partially driven by low net benefits of sustainable forest management. Low prices for timber, in part due to the supply of logs from unsustainable and illegal operations, discourage investments in long-run forest management. So do a lack of credit and secure tenure for private forest lands, or concessions on public lands, with insecurity requiring costly private protection. Services provided by forests often do not generate private revenues and therefore do not affect private landowners' decisions. While this lack of incentives for sustainable forest management mainly drives forest degradation, it also indirectly raises deforestation, as since standing native forest is often left more accessible for clearing, as well as more vulnerable to disturbances such as fire, as a result of both poorly planned and implemented logging and collection of wood fuel.

We consider these sectors using 3 'generic' model types to illustrate incentives relevant for policy impacts on forests. Their microeconomic perspectives are relevant for deciding across possible "nationally appropriate mitigation activities" (NAMA), complementing Kerr and Streck plus Lubowski and Rose in this Symposium which presume local knowledge of efficient NAMA. Angelsen and Rudel in this Symposium ask what policies each specific country can manage and will choose, focusing on differences among countries and the development paths or 'transitions' between countries' distinct settings. That complements our highlighting of incentives issues that are common across space and time for each model, i.e. each decision agent and market structure.

The 1<sup>st</sup> model is "profit maximization by producers who are well integrated in markets". This framework has been applied to many decisions about best locations for profitable land uses; predictions have been widely studied empirically, including using large-scale secondary datasets with considerable variation in travel time to major markets. Our 2<sup>nd</sup> model, which we refer to as "household optimization with both production and consumption motivations", has been applied to more isolated settings featuring transactions costs that generate incomplete market integration. Thus its predictions of policy impact link to variation in household assets and preference and are tested with household and village surveys. Our third model, labeled "public optimization", aims to integrate elements of literature on concessions and corruption, focusing on forest degradation and with immediate implications for decentralization as parameters vary with government scale. Predictions can be tested using observed policy, while impact can be examined at various scales.

Our focus is the incentives underlying whether or not policies affect land use and forest. We review the existing empirical impact analyses for both 'forest' or 'conservation' policies, i.e. domestic efforts to impede the clearing or degradation of tropical forest, and shifts in key driving forces, including 'non-forest' or 'development' policies. Changes in either type of policy could help a forested country achieve commitments under REDD and be part of the domestic NAMA.

We find that incentives driving forest loss often were not addressed in past policy design. That both helps to explain limited previous impacts of policy and suggests shifts in future policy. It will be critical to base policy on the relevant model for a given setting, as shifting any specific incentive may reduce forest loss in one setting but not another. In sum, both theory and evidence suggest that REDD+ policy success will rely upon our understanding of the full set of incentives. While respecting the potential complexity that implies, we end by drawing on past experiences to suggest ideas for increasing policy's reduction of emissions from deforestation and degradation.

Below, Section 2 gives a brief complete-market model of producer profit maximization and our review of empirical evidence for a suite of policies for which this generates predictions. Section 3 presents a model of household optimization given incomplete markets and a review of the empirical evidence which focuses on how and when that incompleteness shifts policy impact. Section 4 provides a summary model integrating key pieces of public optimizations concerning concessions and corruption, with a focus on degradation and implications for decentralization. Finally, Section 5 looks forward at REDD+ policy choice in light of the findings in the review.

#### 2. Producer Profit Maximization

#### 2.1 Simple Core Model

Tropical deforestation is driven by demand for land in agriculture, with few exceptions. Thus, changes that influence agricultural profit from cleared land, relative to profits from forest management on that same land, will affect the rate of deforestation. Some such changes in social and economic settings certainly affect forest but yet are highly unlikely to be adopted as REDD policy, including recessions, currency revaluation, and violent conflict (Rudel 2000, Lopez and Galinato 2005, Morton et al. 2008). Other policies that influence relative profits, though, might shift if global REDD payments make reduced loss of tropical forests into a domestic policy goal.

As a first step in assessing whether and when domestic policies might reduce forest loss, i.e. produce REDD, we present briefly the very standard model of producer profit maximization integrated with perfect markets which has so often been the basis for considering policy impacts. This framework is implicit in predictions and empirical examinations of land use, and forest loss, in many literatures. Once we have laid out its elements, we draw connections to various policies.

For any forested site the producer chooses the use of the land, in forested or cleared state, to maximize the expected profit. Expected revenue from that site is affected by the output prices, transport and other costs and a risk of expropriation of land in forest if forest tenure is not secure. The cost of producing any unit of output is affected by the prices of labor, capital and land and by the productivity of land. Sites' heterogeneity is often emphasized in this framework, i.e. one important feature is significant spatial variation in profits from agriculture or forest management. Many have applied these ideas in varied ways, so we keep our 'generic' summary model simple:

(1) maximize<sub>{D}</sub> site profit<sub>i</sub> = D •  $\pi_{clear}(p,\tau,d_i,c_i,\varepsilon_i) + (1-D) \cdot (1-r_i) \cdot \pi_{forest}(p,\tau,d_i,c_i,\varepsilon_i)$ where:

D	= 1 if deforested or degraded, 0 if managed for forest output
р	= prices of outputs at market, including the price of timber
τ	= transport cost per unit distance to market
$d_i$	= distance from the market to the site
$C_i$	= cost of producing a unit of output given rain, slope, soil
$\mathcal{E}_i$	= privately (or locally) observed factors in profit (with $cdf F$ )
<i>r</i> <sub>i</sub>	= risk of public expropriation or private trespass, if still in forest

This can be applied to a producer's decision to clear any given site. Across the landscape, this is just a form of the familiar von Thunen prediction. Fixing *c* and  $\varepsilon$ , if net agricultural profit is positive at *d* = 0 and falls with *d<sub>i</sub>* there is clearing up to a given market distance, beyond which forest stands (Ehui and Hertel 1989, Barbier and Burgess 1996). Allowing spatially stochastic  $\varepsilon_i$  to vary for any given market distance, the probability of deforestation falls with market distance.

The above framework is relevant for predicting the forest impact of a large set of policies. Tenure regimes where title is easier to obtain and defend after clearing (e.g., r = 0 if D=1 above) will increase the rate of forest clearing, as would supports that raise agricultural p and/or lower c. Deforestation avoided by perfectly enforced protected areas is lower far from market (higher  $d_i$ ). REDD payments for carbon storage (p > 0 if D=0) can directly encourage additional forest and, for binary clearing choice, also reduce the marginal clearing impact of changes in other factors.

Of course, not only unobservable quality  $\varepsilon$  but also observable costs c vary considerably. Consequent spatial variation in expected profits implies spatial heterogeneity of policy impacts. Consider, for instance, the paving of an unpaved road, which lowers  $\tau$  from the market to a site. This increases profit and should affect land use at *marginal sites* which would almost have been worth clearing with just the unpaved road. Again fixing c and  $\varepsilon$ , to consider this most intuitively, these are the sites just beyond the prior market-distance threshold for clearing of the forest. As in fact the  $\varepsilon$  varies for any given distance  $d_i$ , impacts vary as a function of the probability density f. If  $\varepsilon$  is normally distributed about zero then impact is highest at the threshold (Pfaff et al. 2011a) and new roads (lower  $\tau$ ) deforest less if they go where old roads already led to clearing (low  $d_i$ ).

It is worth highlighting a wide range of choices (beyond (1)) to which such logic applies, from allocations between clearing and forest to choice about management such as the use of fire. Responses to tenure insecurity include site protection and less investment in future forest stocks, at least when tenure has been less secure for forest, such as when clearing is required for tenure. For PES programs, producers decide whether to participate and, if so, which lands to volunteer. Within a concession, a firm chooses how and how much to harvest, plus whether to fully report. Monitoring of clearing, and of corruption, affects whether to violate agreements or to pay bribes.

Finally, note that  $\varepsilon$  is not observed by an analyst or a federal agency located in the capital. Yet at least some of private information  $\varepsilon$  is likely to be observed by local government, so federal 'one-size-fits-all' policies based upon observables can be less efficient than local policy choices.

#### 2.2 <u>Reviewing Empirical Evidence</u>

# 2.2.1 Development ('non-forest') Policies

We consider policies not aimed at forests but still relevant due to their impacts upon agriculture. These 'non-forest' or typical 'development' policies significantly influence the clearing pressure that must be addressed by any conservation policy. They could be shifted to reduce forest losses, if that becomes a domestic goal as a result of global payments based upon REDD measurements. Following Kerr and Streck in this Symposium, we note that without an explicit contract based on REDD measures, still global funds could cover the cost of shifts in development policies that are believed to generate REDD -- very much in the spirit of GEF funding for 'incremental costs', i.e. the marginal cost incurred by a forested country of a policy shift that generates global ecobenefit.

#### Transport / Infrastructure / Development

New road investments improve access and lower commodity transport costs and thus can lead to greater forest loss (a rise in forest-management profit would have a countervailing effect). This general prediction from (1) is supported by analyses of broad-scale, secondary data on the biogeophysical and socioeconomic factors in profits (Chomitz and Gray 1996, Kaimowitz and Angelsen 1998, Pfaff 1999, Geist and Lambin 2002, Pfaff et al. 2007). While government roads open new regions, logging operations greatly expand access by building unofficial roads (for the Brazilian Amazon Veríssimo et al. 2002, Brandão and Souza 2006, Perz et al. 2007; for Congo basin, Laporte et al. 2007). REDD payments can lead to reconsideration of whether and where governments should build roads and of whether to allow new private roads (Arima et al. 2008).

Applying (1)'s logic to new road investments given prior road infrastructure suggests that new road impacts vary over space so the spatial structure of development affects forest loss rates. Nelson and Hellerstein 1997 find that the existence of a prior road affects impacts of new roads and Andersen et al. 2002 find that higher prior deforestation lowers forest impact of new roads. Pfaff et al. 2011a,b show that (1) predicts low short-run impacts also within isolated regions and confirms these predictions looking across the Brazilian Amazon (see also Conde and Pfaff 2008 and Delgado at al. 2010). Given spatial variation in the forest impacts of roads, REDD payments could reward 'clean' locations. Angelsen and Rudel in this Symposium suggest that shifting new roads from frontiers to 'mosaic' developed settings could increase both REDD and rural welfare.

These implications of profit maximization from (1) apply to other infrastructure impacts. Governments also boost local profits with agricultural credit infrastructure and support for small-farmer settlements near forest (Fearnside 2005c, Barreto 2008) that can be spatially targeted too. Energy pipelines are sure to increase within the western Amazon – within and outside of Brazil – and Finer et al. (2008) show that the many 'blocks' newly zoned for energy development overlap with species-rich forest. We note that pipeline development often is accompanied by new roads, for construction and maintenance. They could follow and extend prior paths or create new paths; again the spatial structure of these and other development policies is likely to affect forest losses. Finally, another implication of (1) is that not only prior roads but also other policies influence the marginal impacts of roads. Sills et al. (2006) describe Acre State in the Brazilian Amazon, where policies including protection, payments and monitoring aim to lower road impacts on the rate of deforestation while still harnessing benefits of market access for sustainable forest management.

#### Agriculture

Considering different elements of profit in (1), governments subsidize agriculture through support for higher output prices (e.g., using import tariffs upon competing products, subsidized processing) and through support for lower input prices (such as interest rates and fertilizer costs) as well as through lower taxes. Governments also help to reduce pests and disease risks through investments in research and development. Without off-setting public support for forested lands, such policies that raise yield and profit in agricultural land uses will generate more loss of forest. Such ties are well known, longstanding features of policy in developed and developing countries. Yet such policies and relationships may be reconsidered under REDD, especially given domestic constituencies (Keohane and Levy 1996 and Ross 1996 stress the local stakeholders' interests).

Biofuels merit mention given recently increased attention to, and action within, that area. If they are cultivated on croplands, profit maximization can imply a rise in forest clearing as the supply of the displaced agricultural commodity (e.g. soy) will fall and thus its price will increase. Biofuels subsidies on land not currently in agricultural production might lower carbon emissions from the production of fuel without more forest loss, though it may not be very productive land.

#### Tenure / Security

As in (1), clearing is promoted if it yields squatter's rights and eventually title (Fearnside 2005a; Simmons et al. 2009 on 'direct action land reform' note similar and ongoing incentives). Even clearing that is not followed directly by a profitable land use may allow acquisition of title, facilitating credit for future profitable production as well as permitting a future profitable resale.

Incomplete and overlapping records that do not recognize traditional land tenure permit powerful actors to acquire titles to vast tracks of land (Fearnside 2005a; Brito and Barreto 2009) Such risks to one's land, from either public expropriation or private trespass (as noted in (1)) can reduce the incentive for long-term, sustainable management of forested land (Araujo et al. 2009). Exploitation by illegal loggers and trespassers is widespread in tropical forest regions (Armsberg 1998, van Kooten et al. 1999, Blaser and Douglas 2000, Bohn and Deacon 2000, Contreras-Hermosilla 2000, Putz et al. 2000, Guertin 2003, Cerutti and Tacconi 2008, Honey-Rosés 2009). Alston et al. 2000 argue that in the Amazon, in the past, only land clearing for at least five years protected against expropriation risk. This risk, and clearing's key role, promoted deforestation.

Yet security could increase profitability of clearing, for instance investments in clearing for oil palm. In general, security raises the weight on long-term payoffs (Mertenes et al. 2002) and may or may not support sustainable timber management. Its effect also can be changed by household differences as in (2) below but even in (1) its impact depends on the suite of payoffs.

Another profit maximizing reaction to insecurity is to physically defend one's land rights. This is costly and likely cheaper in agriculture (Armsberg 1998), influencing choice of land use. Such costs are key if government enforcement is poor, due to budgets, remoteness or corruption and will reduce investment in intensive sustainable forestry (e.g., plantations) as well as in other forest capital (see Wibowo and Byron 1999, Bohn and Deacon 2000, Barbier and Burgess 2001).

Finally, security can affect profit-maximizing migration and thus forest when agricultural frontiers spill into forests (Perz et al. 2005 correlate rural population growth with deforestation).

In tropical forest areas in the Americas, fertility rates are high but in-migration is the main driver of population growth and both are partly driven by tenure insecurity (Carr 2004; Carr et al. 2006; Bilsborrow 2002). Others identify a lack of tenure as a "pull" as people go where access is clear but tenure not, so as to obtain plots (Amacher et al. 2008, Barnes et al. 2002, Merry et al. 2008).

#### 2.2.2 Conservation ('forest') Policies

#### Protected Areas (PAs)

PAs have been the most common policy for explicitly targeting forest conservation. They have grown substantially over recent decades. They are also close in spirit to carbon-storage set-asides some have envisioned for REDD (Brandon and Wells 2009). To this point, protection has gone to sites with relatively low current threats (Joppa and Pfaff 2009 for global documentation) and thus it has reduced deforestation much less within park borders than seems to be presumed. This follows from (1) given that, for whatever reasons, on average PAs have avoided pressures. While isolated PAs could be optimal, including due to the cost of acquiring and protecting land, far from roads and from cities and on high slopes and poor soil one might expect standing forest even in the absence of protection (as in (1)), so even a fully forested PA may not indicate impact.

Joppa and Pfaff 2010a highlight the lack of application of (1) in construction of baselines for impact evaluation. Many use clearing on all unprotected sites to estimate the counterfactual. Others use the clearing of a 'spatial buffer', using proximity as a proxy for measured similarity; this could work but may not, especially if boundaries are drawn using geographic distinctions, for example drawing the boundary where slope starts to rise or on the far side of a broad river. Andam et al. 2008 for Costa Rican PAs use matching on site characteristics and find an impact under half the other two estimates for 1963-1997. Pfaff et al. 2009 confirm this average impact and test the implication from (1) that PA impacts in a given country can vary a lot across space while global results in Joppa and Pfaff 2010b, for over 100 countries, support both conclusions. Ongoing Brazilian Amazon work confirms both too (Delgado et al. 2008, Pfaff et al. 2011c,d,e).

Finally, we consider implications of (1)'s profit-maximizing reactions *outside* of the PAs. PA spillovers have implications for net forest impacts and socioeconomic impacts, e.g. lowering the labor demand within agriculture (Robalino 2006) or raising it in tourism (Andam et al. 2010). The sign of net spillovers from PAs is also unclear for forest. In Costa Rica, Robalino et al. 2011 find higher clearing in spatial buffers (specifically in subsets where spillovers are most expected) than for matched controls whereas for the Brazilian Amazon, Pfaff et al. 2011f find less clearing.

Different spillovers are consistent with (1), given that the two are very different settings. That serves to nicely highlight the differences between (1) and (2), i.e. our second model below, and links to Angelsen and Rudel in this Symposium, which is focused precisely upon differences in countries' settings and the transitions between those settings. Our second model is for settings that are more isolated, on the forested frontier, where various markets are not always integrated; that describes the Brazilian Amazon much better than Costa Rica. Also, in such frontier settings public optimization (as in our model (3)) interacts with private within the development dynamic. Combining perspectives, Pfaff et al. 2011f suggests that within such a dynamic public protection could signal a lack of future public development infrastructure and thus influence private actions. This creates an additional channel, not present in the Costa Rican setting, to lower loss of forest.

#### Payments for Ecosystem Services (PES)

Payments for ecosystem services such as water quality, species habitat, or carbon storage can reward owners for conserving the forest, instead of clearing (by creating p>0 for D=0 in (1)). While services generated by payments are likely to benefit those who are paying, gains in forest stocks could also benefit those who are collecting non-timber forest products including fuelwood (Kohlin and Amacher 2005, Arnold et al. 2006, Cooke et al. 2008) and thereby also be providing some type of 'natural insurance' in the face of varied income shocks (Pattanayak and Sills 2001).

We emphasize why payments may not have significant impacts. Two obvious reasons are that they might be too low to sway the decision, especially if aiming for very high pressure lands, and that if like PAs they go to relatively low-pressure lands that could reduce payments' impacts. Additionally, though, PES programs tend to involve landowners volunteering the land and for an identical payment (1) suggests lands with low profit, and thus less clearing threat, may be offered (this and other relevant points on PES were made previously in REEP in Pattanayak et al. 2010).

For early Costa Rican PES, participants differ significantly from others in characteristics that affect land use (Ortiz et al. 2003; Miranda et al. 2003; Zbinden and Lee 2005); clearing may be lower on paid parcels for reasons unrelated to payments (de Camino et al. 2000; Brockett and Gottfried 2002; Miranda et al. 2003; Sánchez-Azofeifa et al. 2007; Arriagada et al. 2009). Yet a larger constraint on impact was a low national clearing rate. Not only the recent matching studies (Arriagada et al. 2011, Robalino and Pfaff 2011) but also prior work (Sierra and Russman 2006) find that a great majority of the paid land would have remained in forest even without payments.

Still, it is worth understanding what if any conditions and rules for allocation of payment generated higher (albeit still low) impacts than average. Arriagada et al. 2011 focus on a location where a NGO played a strong role in targeting threats; this seems to have clearly raised impacts. Robalino et al. 2008 focus on how PES selection rules varied across time and space after 2000.

In Mexico, early payments were for hydroservices and it is suggested (Munoz et al. 2008) that they went to sites facing relatively low threats. Again, even without knowing why this arose, our model (1) implies that the impacts are somewhat constrained conditional on those locations. A recent investigation of these payments used matched controls from the set of volunteers to set the counterfactual baseline, i.e. the estimate of what would have occurred without the payments (Alix-Garcia et al. 2011a). That nicely controls for some of the  $\varepsilon$ , i.e. private information in (1), that likely differs between volunteered and not. Consistent with greater national clearing rates in Mexico, they find statistically significant and small-to-moderate avoided-deforestation impacts.

Finally, just as for PA spillovers, the full set of the profit-maximizing responses by actors to having payments can imply impacts *outside* of the paid parcels. The same recent Mexico study tested for two types of slippage, or leakage: increased deforestation on other property belonging to program recipients (substitution slippage); and increased deforestation within markets where there are high levels of program participation (output-price slippage). Much as for PA spillovers, the setting in which these payments were made appears to affect the magnitude of the spillovers. In the Mexico study, output-price slippage was significant when relatively isolated from markets, i.e. the case when we would expect the demand for outputs to be addressed through local supply. That motivates model (2) below and links to themes of Angelsen and Rudel in this Symposium.

#### 3. Household Optimization Given Incomplete Markets

Where markets are complete as in (1), and all above, deforestation often is modeled with careful empirical attention to spatial heterogeneity but less attention to heterogeneity across producers (Mertens et al. 2002 bridges (1) and (2), applying a model like (1) to different household types). But on forest margins, farmers typically are not fully integrated with some markets -- often rural households whose income is farm profit and who consume goods similar to those they produce.

These two features of forest margins, incomplete markets and integration of production and consumption, can affect REDD+ by changing policy impacts and creating new policy levers. They imply that the household's endowments and preferences influence production choices like the clearing of land and harvest of forest products, through impacts on households' incomes and marginal values of inputs such as labor and land. Thus, heterogeneity of household producers is often a central feature in models, while spatial heterogeneity is limited by geographic constraints upon the household-level surveys that often provide the data for empirical testing of predictions.

### 3.1 Conceptual Shifts Relative To Producer Profit Maximization

On forest margins, high transaction costs yield incomplete or effectively missing markets. Most rural economies are thought to be characterized by incomplete markets for both labor and land (Sonoda and Maruyama 1999, Carter and Yao 2002). Limited access to formal credit and insurance plus limits on informal mechanisms for addressing contingencies within communities (Besley 1995, Barrett 1999, Bhattamishra and Barrett 2010) as well as shallow markets for forest products are also the norm in many tropical forest regions (Shanley et al. 2002, Sills et al. 2011).

Household-production models have been widely applied to the agricultural and the forest production behavior of rural households. Theoretical models typically focus on labor allocations, e.g. between clearing of forest, agricultural production on cleared land, harvest of forest product, domestic chores and wage employment. Allocation to clearing and harvesting of forest products are of interest for REDD+. They are influenced by biophysical characteristics and market access, such as highlighted in (1), but also endowments and preferences because markets are incomplete.

Much of the intuition here can be captured by conceiving of market incompleteness as the existence of 'price bands' between the purchase prices and the sale prices for outputs and inputs (Sadoulet and de Janvry 1995, Pendleton and Howe 2002, Sills et al. 2003). Thus, a household pays more to purchase a good than it receives if selling the same good. This could be because of transportation costs, e.g. not only a fixed cost of getting oneself to market but also a unit cost of getting one's goods there or inputs back to the farm, or information costs, for instance a lack of market experience or communications making it difficult to find the best purchase or sale price.

These costs depend on general conditions like quality of roads and cell-phone networks, but also household characteristics like numeracy or vehicle ownership. Thus, the price band for outputs and inputs including labor varies by household. Depending on the internal valuations of goods and labor relative to market prices, a household might produce primarily for consumption and/or mostly or only use family labor for on-farm production, instead of hiring or selling labor.

Thus households in such settings may respond differently to shifts in price or technology. As in (1), higher agricultural profits encourage households to deforest more land for production.

But when clearing uses household labor, in the style of backward-bending labor-supply curves a rise in labor's value marginal product and household income can raise leisure demand, reducing time allocated to clearing. That could offset some or all of the direct price or technology effect (more dramatically so if households satisfice rather than maximizing utility, as in the "full belly" model Angelsen 1999 argued was implicit in much of the policy discussion over deforestation).

More generally, market imperfections (e.g., price bands) can influence the net effects of shifts in relevant prices and production technologies (Angelsen 1999, Pendleton and Howe 2002, van Soest et al. 2002, Muller and Albers 2004, Caviglia-Harris and Sills 2005, Takasaki 2007) by creating roles for household preferences and endowments (see Walker et al. 2002, Caldas et al. 2007, and de Sherbinin et al. 2008). These additional channels for varied behavioral impacts can affect net impacts of REDD+ policies, or pilots, that change conditions for rural households: dissemination of new agricultural technologies; clarification of tenure; improvements in access to credit; adjustments to prices; creation of employment in tourism or processing forest products; and influences on household preferences across farming, forest harvest, and off-farm production.

Another important example is localized markets, where the imperfection is broader scale. Households may be integrated in local markets that at larger scales are isolated by transport cost. This intermediate case underlies, for instance, Alix-Garcia et al. 2011a's PES spillovers (2.2.2). As in (1), PES raise opportunity costs of clearing forest for producers and can reduce forest loss. But shifts in preferred household consumption given the increase in income from these payments can matter (Alix-Garcia et al. 2011b find such impact from a conditional cash transfer program). With the payment income in hand, a household may prefer to consume more of what is produced through local clearing of forest, for instance more beef or milk. This can encourage both leakage, e.g. keeping the enrolled land in forest to earn the ecopayments while clearing other forest lands, or simple violation of payment terms by clearing enrolled land (cf., Muller and Albers 2004).<sup>i</sup>

This intuition can be formalized in household production models (see Singh et. al 1986). Variants on (2) below have been used to describe farm households on forest margins who are one key group for reducing deforestation, e.g. colonists in the Amazon and farmers in Southeast Asia who utilize lowland and/or upland farming systems. Such a modeling framework has also been applied to interventions that seek to promote forest use as a sustainable alternative (see Weber et al. 2011 and Sills et al. 2003) or to deter collection of forest products that leads to degradation (e.g., Muller and Albers 2004, Fisher et al. 2005, Amacher et al. 1999, and Baland et al. 2010).

Following Sills et al. (2003) and Fisher et al. (2005), consider a typical rural household on the forest margin maximizing utility from consumption purchased ( $q_o$ ) and produced ( $q_a$ ,  $q_f$ ) plus home time ( $t_H$ ) given heterogeneous preferences ( $\Phi$ ) and technologies or fixed inputs ( $\Psi_{a,f}$ ).<sup>ii</sup> Households produce agriculture ( $Q_a$ ) and forest products ( $Q_f$ ) with purchased inputs ( $i_c$  at cost c) and hired time in clearing ( $t_w$ ), where effort can be observed, by allocating their time (T) among: home time; agriculture ( $t_a$ ); clearing for agriculture ( $t_D+t_w$ ); gathering of forest products ( $t_f$ ); and off-farm wage labor ( $t_o$ >0). A common variation is to include the time required to walk to work, based on distance from the household residence to the agricultural fields or the forest (Angelsen 1999, Brown 2008, Maertens et al. 2006, Fisher et al. 2005, Robinson et al. 2002), which yields local landscape predictions similar to 'von Thünen patterns' around markets as described in (1). Model (2) below makes some particular choices to convey basic intuitions (also see endnotes): (2) maximize<sub>{q, t}</sub> U (q<sub>o</sub>, q<sub>a</sub>, q<sub>f</sub>, t<sub>H</sub>;  $\Phi$ )

subject to these constraints:

$T + t_w \ge t_H + t_a + t_D + t_f + t_o$	Household Time Constraint
$Q_f = f(t_f, L_D; \Psi_f)$	Forest Product Collection
$Q_a = a(i_c, t_a, L, L_D(t_D, t_w); \Psi_a)$	Agricultural Production
$I(L_D) + \Sigma_{i=a,f} (p_i - \xi_i) \bullet Q_i + (w - \xi_t) \bullet t_o \geq p_o q_o + p_a q_a + p_f q_f + ci_c - c_i + $	+ wt <sub>w</sub> Household Budget

Forest production (f) rises with time collecting and with deforestation because slash-andburn results in both cleared agricultural land and forest products, e.g. timber or fuelwood (Fisher et al. 2005). We model collection in abundant open-access forest, not diminished by agriculture. Forest output is affected by endowments such as human capital and biophysical conditions ( $\Psi_f$ ).<sup>iii</sup>

Agricultural output rises with time in agriculture ( $t_a$ ) and own and hired clearing ( $t_D$ ,  $t_w$ ), given fixed household production endowments and technologies, including human capital ( $\Psi_a$ ). Land may have been cleared in the past (L) and may be cleared in the period that we model ( $L_D$ ). That is, in this one-period model, land is cleared and farmed in the same period ( $L_D$  adds to  $Q_A$ ).<sup>iv</sup>

The budget requires that expenditures on consumption  $(q_i)$ , inputs  $(i_c)$  and labor  $(t_w)$  be no greater than the value of production plus labor earnings  $(t_o)$ , unless net external income (I) from remittances, social welfare programs or REDD payments, e.g., permit additional consumption of goods or home time. To focus on REDD+, income I depends on L<sub>D</sub> to represent a transfer that is conditional on not clearing land.<sup>v</sup> Price bands for goods and labor are per-unit transaction costs  $(\xi_a, \xi_b, \xi_t)$  that could be transport cost or could be the losses from imperfect market information.<sup>vi</sup>

#### 3.2 <u>Reviewing Empirical Evidence</u>

#### 3.2.1 Development ('non-forest') Policies -- Additional Household Perspectives

#### Transport, Transitions & Development Policies

Travel cost from the household residence to a parcel, or the nearest road or market center, is a very robust predictor of deforestation. While household endowments and preferences matter, empirically they do not swamp the effects of market access. This is confirmed at household level by linking remote sensing data on deforestation with agent surveys (Fox et al. 2003, Verburg et al. 2004, Sills and Caviglia-Harris 2008, Caviglia-Harris et al. 2009), with Maertens et al. 2006 providing a story, based on issues raised in (2), for heterogeneous road impacts (see also 2.2.2).

This may seem inconsistent with our focus here upon more isolated settings with market incompleteness. However, that focus actually highlights the key role of transport / infrastructure investments in integrating such areas more fully with the markets. Thus, road construction may effectively move a region from our model (2) setting into our model (1) setting, emphasizing the dynamics of development paths that are the focus of the Angelsen and Rudel Symposium article (and Tachibana et al. 2001 find impacts shift with development and integration of labor markets while Pfaff and Walker 2010 and Meyroidt et al. 2010 highlight transport's roles in goods trade, which can bring great change in one locale that relies on opposite changes within other locales).

Yet in (2), economic integration may expand off-farm labor opportunities (Barrett 1999) that increase households' incomes (Zwane 2007). Thus, at first integration via new roads might raise pressures for clearing but then if labor shifts off the farm pressure could fall again, e.g. an 'environmental Kuznets curve' for household land use (Atmadja and Sills 2008). Shively 2001 finds reduced labor allocation to farming as hiring in falls with higher wages while the hiring out of family labor rises; improved integration may yield migration towards external labor demands. Yet if migrants are pushed out of home regions into frontier subsistence settings, where income effects arise with the reliance on household labor, the latter drop in pressure may not materialize. Education that raises opportunity cost of home labor is commonly found to reduce deforestation, offering a long-term development lever if those paying for REDD hope to phase out payments.

Another development policy suggested by the importance of household labor within (2) is family planning (Pan et al. 2007). Household sizes or the adult males often are linked to clearing. In contrast, Van Wey et al. (2007) find the number of women has higher impact on deforestation and land-use choices in their case study region in the Brazilian Amazon. They hypothesize that these effects of household composition on forest are due to off-farm employment opportunities (Barbieri and Carr 2005) as well as social welfare payments to women and girls that relax cash constraints and allow the purchase of inputs (including hired labor) for expanding agriculture.

That raises in turn an increasingly common development policy, conditional cash transfer to provide an incentive for development, e.g. for attending school. As noted above, Alix-Garcia et al. 2011b find for Mexico that an increase in exogenous income from the conditional transfers increases deforestation in local areas not integrated into broader markets and this results is quite consistent with Foster and Rosenzweig 2003's finding of localized impacts of income if markets are incomplete, with positive or negative impacts on forest depending on the income elasticities.

#### Agricultural Policy Impacts Given Household Income & Labor

Model (1) predicts that higher output prices and improvement in agricultural technologies raise clearing. Yet with incomplete markets in (2) shifts in output prices and technology can have income effects that increase leisure's value for goods net sellers, thus reducing a rise in clearing. But for net buyers, higher output prices clearly encourage clearing -- through both substitution effects from (1) and income effects on leisure demand from (2) -- while price variation may lead the risk-averse to produce more of their own food, implying more deforestation (Barrett 1999; many of the poorest rural households in Africa may be net buyers (see also Fisher et al. 2005)). Thus predictions from (2) depend on not only how well but also how households are integrated.

Improvements in agricultural technology that raise labor intensity can compete for labor to clear land, potentially even decreasing deforestation, if labor markets are incomplete as in (2). Takasaki (2007), for example, predicts that increased returns to labor within current agricultural production could draw labor out of clearing land for future sale (land speculation or "sell-out"). Maertens et al. (2006) and Klasen et al. (2010) find more labor-intensive paddy rice production (e.g., terracing investment) in lowland areas draws labor out of deforestation in highland areas.<sup>vii</sup>

Another policy commonly debated is improving access to credit. Zwane (2007) suggests that the impact may depend upon initial household income due to differences across households in the marginal rate of substitution of consumption and leisure. At lower incomes, more clearing

is likely, while at high incomes a rise in demand for leisure could dampen this effect. Zwane also predicts larger households would be more likely to use the credit for extensification, i.e. clearing. It is commonly argued that cattle are a 'walking savings account' if access to credit is limited, so that better credit access could reduce reliance on land-intensive ranching. Yet preferences may drive cattle ownership beyond roles of limited credit access and risk aversion (Merry et al. 2004, Siegmund-Schultze et al. 2007 for the Brazilian Amazon). Preferences may not easily be shifted.

#### Tenure Security -- Household Evidence

In much of the household production literature, the tenure regime is implicit in the model. The most common structure has the household acquiring additional farm land solely through the investment of labor in clearing. This implicitly assumes clearing confers tenure, i.e. zero private property rights to forest but secure rights to cleared land. The degree to which this is true varies.

Thinking on tenure's impacts, often regime does not vary in a household surveys dataset. Deininger and Jin (2006) and Ayalew et al. (2011) for Ethiopia do find insecure tenure decreases tree planting. Shively (1998) shows investments in trees are sensitive to price fluctuations and to risk preferences, with application to the Philippines. Thus, policies that increase wood fuel price (such as reduced access to public forests for charcoal production) and increase security of tenure can be important for raising the supply of wood fuels from private sources (Köhlin et al. 2011). If woodfuel substitutes for fossil fuels, this has implications for total greenhouse-gas emissions. However plantation trees have multiple uses and use as woodfuel may not be the most valuable.

#### 3.2.2 Conservation (Forest) Policies

In conservation applications of (2), again heterogeneity is a big theme for forest margins (see Campbell et al. 2010, Muller and Albers 2004, Blom et al. 2010, Brandon and Wells 2009). Coomes et al. 2004 say a weakness of past conservation-development initiatives "lies in their founding on a limited understanding of the microeconomic logic that gives rise to livelihood heterogeneity among forest peoples". That fits our themes well, although another limit has been a lack of sustained funding or donor attention (Sills et al. 2011, Bauch 2010, Blom et al. 2010). This constrains local participation, hampering 'buy-in' commonly recognized as one key element of success (Brandon and Wells 2009, Brooks et al. 2006). Finally, the lack of careful monitoring and evaluation, including in light of varied actors and goals, limits our ability to generalize and to predict (Brooks et al. 2006, Brandon and Wells 2009, Blom et al. 2010, Jagger et al. 2010).

#### ICDPs (Integrated Conservation and Development Projects)

ICDPs have been the dominant attempt to reconcile conservation with local livelihoods (Brandon and Wells 2009). They focus on increasing local incomes in new economic activities such as ecotourism, beekeeping, or sustainable harvesting and the processing of forest products. Activities are intended to: [i] require intact ecosystems as inputs; [ii] draw labor out of activities that are environmentally harmful; and/or [iii] compensate for and thus encourage acceptance of legal restrictions on previously accessible natural resources (see, e.g., Weber et al. forthcoming).

Concerning [i], the objective is to increase household demand for quality standing forest, in the spirit of (2) as well as incomplete markets focusing on local forest supply. The mechanism is the introduction of a production function that requires the local forests (shadow p>0 for D=0).

If usable forest is limited by the amount of land a household owns, or distance from a residence, then creating new derived demand for forest could increase opportunity costs of clearing forest.

Concerning [ii], the link to (2) and to incomplete labor markets is explicit and immediate. The objective is to increase the demand for household labor or, as in Ferraro and Simpson's 2002 critique of ICDPs, "distract" households from deforestation. Muller and Albers 2004 show this is most likely to have impact if labor markets are incomplete so households use their own labor.

Concerning [iii], the motivation for compensation could simply be ethical but also could leverage local notions of fairness in order to encourage local 'acceptance' of forest conservation. Within (2)'s isolated forest margins, a value of acceptance derives from on limits on monitoring. That buy-in matters accords with the results of behavioral experiments on 'motivation crowding' where attempts to restrict from the outside end up backfiring (for instance Cardenas et al. 2000).

Prior literature reviews have found little evidence of ICDP effectiveness (McShane and Wells 2004, Naughton-Treves et al. 2005, Brooks et al. 2006, Garnett et al. 2007). Reviewing interventions to alleviate poverty while conserving biodiversity, Leisher et al. (2010) do provide some evidence that community forest management for timber<sup>viii</sup>, nature tourism, and mangroves can have significant benefits for both poverty alleviation and conservation. They note, however, that both participation and the benefits often are concentrated among the better-off households.

For forest-based microenterprise in the Tapajós National Forest in the Brazilian Amazon, Weber et al. (forthcoming) and Bauch 2010 find households with more durable assets and less pasture were more likely to participate. Controlling for selection, participation did increase cash income and asset accumulation but there is no evidence of reduction in activity requiring clearing (e.g., growing manioc or raising cattle). This ICDP may not generate enough income, a common issue as on forest margins there may be barriers to NTFP sales (Shanley et al. 2002 and below). Ecotourism has been most successful where charismatic species are a key attraction to tourists and a conservation concern (Krüger 2005), which is often not the case for dense tropical forest. This does not rule out the possibility that this ICDP either reduced forest conflict or increased local acceptance of conservation regulations, both of which are relevant concerns for REDD+.

# NTFPs (Non-Timber Forest Products)

Sills et al. 2003 broadly categorizes a large NTFP literature as about either on low-value, high-volume products (e.g. wood fuel, fodder) or high-value, low-volume products (Brazil nuts, chicle). The former is focused on reducing collection and degradation pressure, e.g. alternative wood fuel sources or alternative fuels, while the latter is focused on NTFPs and incentives (e.g., p > 0 with D=0 in (1)) that raise opportunity costs of labor and land in agricultural production. Recent literature discounts both possibilities, suggesting that fuelwood is not a major driver of forest degradation to start (except charcoal for a market setting more like (1) (Arnold et al. 2006, Ahrends et al. 2010)) and NTFPs have limited ablility to alleviate poverty and to shift incentives (Wunder 2001). The latter can be linked to (2) and isolated forest margins (Shanley et al. 2002).

Research on NTFPs has shifted to another perspective on household utility related to (2), in particular considering the common case of incomplete -- or just missing -- insurance markets. This suggests roles for NTFPs as safety nets. More generally, the focus has shifted to the various

'livelihoods' roles of the NTFPs used for subsistence and traded within local markets and, thus, actively managed in forests (Sills et al. 2011). Local forest access offers "natural insurance" to help buffer income shocks that could increase with climate change (Pattanayak and Sills 2001, Shackleton and Shackleton 2004, Marshall et al. 2006). Fitting (2), there is great heterogeneity in household reliance on forests for insurance (Paumgarten 2005) given varied shocks (Takasaki et al. 2004), incomes alternatives (Fu et al. 2009), human capital (Fisher et al. 2010) and policies (McSweeney 2005). Klasen et al. (2010) find clearing of forests to smooth consumption in the face of shocks. While the 'livelihood' roles of forest are clearly important for many households, though, they seem unlikely to serve as the primary incentive for conservation in REDD+ policy.

# Fuelwood Collection Restrictions

Restrictions on fuelwood collection (many in South Asia) historically were motivated to protect forest health and to ensure the potential for ongoing timber production in public forests. Recently, 'community' or 'joint' forest management initiatives also have effectively restricted collection (Cooke et al. 2008, Robison and Lokina 2011). Looking ahead, in light of REDD+, some commentators focused on livelihood goals fear that private forest owners seeking REDD funds also will impose restrictions to avoid the emissions from degradation. In evaluating what impact such an intervention might have, it is important to consider baseline fuelwood collection.

Consistent with household time scarcity, as in (2), some household models assume that households collect the bare minimum for daily use. For a range of market and shadow prices for fuel and labor, households who collect may shift to purchasing fuelwood as relative prices shift. Thus restricting access could impose costs but not reduce total collection. Empirically, there is evidence that households do lower fuelwood consumption as market or shadow prices increase but not a lot (Cooke et al. 2008, Hyde and Köhlin 2000, Heltberg et al. 2000, Linde-Rahr 2003, Pattanayak et al. 2004, Van't Veld et al. 2006). In South Asia, where much of the research about fuelwood collection has been conducted, households spend more time collecting if wood is more costly in terms of market price (Amacher et al., 1999), shadow price (Cooke 1998a) or in terms of the forest stock or accessibility (Amacher et al. 1993, Köhlin 1998 and Heltberg et al. 2000).

Some evidence, especially from African settings, suggests that households also respond to fuelwood scarcity with increased tree planting and more reliance on private fuelwood supplies (Arnold et al. 2006, Van't Veld et al. 2006). In the Western Ghats of India, Sills et al. 2003 find that determinants of collection vary across private and public access forests and that ownership of fuel substitutes and livestock substantially reduces wood collection from private access forest. That is consistent with the premise that such private forest resources are treated with greater care. For REDD+, we must consider all likely household responses and all the costs households bear.

# 4. Public Optimization & Forest Degradation

REDD+ payments will most likely go to governments. This could be irrelevant if incentives were passed on to agents as in (1) and (2) but effectiveness will depend on government policy choices. Further, global payments could also target the behavior of government itself, e.g. be conditional upon changes in public policy choices (roads, tenure, PAs) or address uses of public forest lands. Within the tropics, most forests continue to be officially owned by some level of government and they are largely managed through the design, agreement on, and implementation of concessions.

For concessions, and generally, "big" actors such as logging firms, commercial ranchers and plantation companies may not accept a public determination of parameters, as in (1) and (2). They may engage with the public sector to influence those parameters, including via corruption. Both the default parameters and such interactions are likely to vary with the level of government, suggesting that REDD+ impacts should vary with decentralization and private-state interactions. We focus in (3) below on degradation impacts of concessions, corruption, and decentralization.

#### 4.1 Summarizing Theoretical Perspectives

In the large native tropical forests of Amazonia, Western and Central Africa and in parts of Asia, logging is often within concessions allocated by government owners of forest. Given their scope, we might conceive of REDD as 'carbon concessions' which mirror forest-harvesting concessions (for example, there are now ecosystem-restoration concessions in Indonesia). REDD+ should be informed by literature on the design of forest policy instruments including terms of concessions.

While the specifics differ by country, concessions are contracts between the government and a forest harvester, usually awarded through bidding. Designating a specific volume, or area for harvest, these contracts cover small or large areas and can be short or long term (Gray 2002). Firms winning harvest rights pay royalties, lump sum or based upon areas or volumes or species. Contracts have more recently also specified environmental sensitivity including the preservation of certain species, selection to prevent 'high grading' and minimizing damage to the environment (or 'reducing impact'). Public rent capture from these royalties often has been less than adequate. In most cases, the harvesters bidding for concession rights are large firms with enough capital to undertake the concession management obligations and, perhaps, also enough political capital to influence government choice, especially in corrupt administrations (Merry and Amacher 2005).

Nearly all REDD-relevant concessions research has been theoretical as obtaining detailed data on many concessions is difficult, especially on illegalities. In models, governments act first, considering firm response in terms of harvest and employing environmentally sensitive methods. In reality, higher government levels (political layers) determine concession number and location, firms' eligibility (e.g. domestic v. international) and allocation mechanisms (sometimes bidding). Lower government levels choose concession size and length (by zone), taxes, and types and rates of royalties. Models aim to predict forest-user response. Within a typical model, the government maximizes social welfare, a combination of public goods from forests (g below) and the rents to harvesters ( $\pi$ ) and others affected by concessions. In maximizing such a function, like V in (3), the government is assumed to face any (or many) of several possible constraints on its choices. Most common is a binding budget target, like B in (3), that depends upon harvesters' choices.

The harvester chooses harvest level (*l*) and method (*h*), affecting royalties paid (given R, the government royalty function) and public goods from forest g(l,h). The harvester also can pay bribes (*b<sub>i</sub>*) and/or can report harvest levels (*x*) below the actual levels (*l*) implying illegal logging. By nature, (**3**) is a two-player strategic model, as the best public choice is a function of expected harvester response which is a function of both observed and expected future government choices. For example, given effective bribery of inspectors, higher government may not monitor ( $m^*=0$ ). However, the determination of optimal bribes is strategically complex. Were higher government to spend a lot upon 'corruption proofing' of officials with high salaries and also high monitoring, optimal bribes could be zero, as could be the optimal deviations of reported harvests from actual.

(3)	maximize $_{\{m,k1,k2\}}$	;R(.);F(.)}	E[V]	=	$g(l,h) + \beta_1 \cdot \pi$
	s.t. where:		В	<	$I(g) + R(x) + m \cdot (1-\beta_2) \cdot F(x-l,h) - e(m,k_1,k_2)$
			harves	ster infl	uence $\equiv \beta_1 = \beta_0 + b_1 \cdot (1 - k_1)$ and $\beta_2 = b_2 \cdot (1 - k_2)$
			harves	ster pro	fit $\equiv \pi = p \cdot l \cdot c \cdot h \cdot R(x) \cdot m \cdot (1 - \beta_2) \cdot F(x - l, h)$
		g(.)	= pub	lic goo	ds (falls with area $l$ , rises with method $h$ )
	Concessions Parameters Corruption Parameters	$\beta_1$	= rela	tive we	eight on harvester profit ( $\beta_0 \ge 0$ is default)
		р	= pric	e of the	e concession output received by harvester
		С	$= \cos \theta$	ts of the	e more environmentally sensitive methods
		Ι	= exte	ernal in	come, perhaps $I(g)$ with REDD payments
		l	= har	vest lev	el, chosen by the harvester (with $h$ and $x$ )
		h	$= \cos \theta$	tly harv	rest methods to raise $g$ ('reduced impact')
		<i>R</i> (.)	= roya	alty rev	enue function, harvester pays government
		x	= repo	orted ha	arvest level, equal to or less than the level
		F(.)	= fine	e paid if	$\sum x < l$ or if methods deviate ( $h < agreed h$ )
		т	= pro	bability	that any illegal logging will be detected
		$b_1$	= leve	el of bri	bery at the political layer of government
		$b_2$	= leve	el of bri	bery (0-1) at lower layer of government
		$k_1$	= brib	e effec	t at political layer ( $k_1$ =1 guarantees none)
		$k_2$	= brib	e effec	t at lower layers (k <sub>2</sub> =1 guarantees none)
		e (.)	= pub	lic enfo	preement costs, rising in <i>m</i> and $k_1$ and $k_2$

Boscolo and Vincent 2007 and Amacher et al. 2007 find that a revenue-neutral (given B) steeper royalty function R decreases harvest volume (*l*). The B constraint yields less-than-perfect enforcement ( $m^* < 1$ ); governments far from forests choose not to detect illegal logging (low *m*) and harvesters may then under-report (x < l), from (1) if gains exceed expected fines ( $m \cdot F(x, l)$ ).

REDD payment conceivably could be conditional (I(g)), helping to meet budget target B. That could lower pressures to permit more harvest in order to raise revenues through royalties *R*. Also it could permit more enforcement (*m*) that lowers illegal logging (Amacher et al. 2008) and, at a cost of internal process (*e*), could lower the influence of bribes on public choice (raise  $k_1$ ,  $k_2$ ).

Illegal logging is always a problem within concessions. Boscolo and Vincent suggest that with low enforcement (*m*), there is little impact of the royalty rates (*R*(.)) upon logging practice<sup>ix</sup>, consistent with results in Clarke et al. 1993. Amacher et al. 2008 find, in two different systems, that under-declaring harvest volume (x < l) rises with royalties (*R*) but falls with enforcement (*e*).

Bribery enters (3) in two ways. At higher layers  $(b_1)$ , it raises the social weight upon  $\pi(.)$ , potentially influencing high-level policy. Bribes to inspectors  $(b_2)$  to overlook any illegal logging changes overall detection success even when the government is seeking to punish bribery. A key feature in the impact of bribes may be inspectors' wages (Jain 2001, Amacher et al. 2011). Thus, the government might pay inspectors more (e) to counteract such low-level bribes (by raising  $k_2$ ). Nonetheless, clearly any influential bribes could hinder the implementation of REDD+ policies.

Decentralization is implicitly modeled as different governments face variations upon (3) including varied weight on g versus  $\pi$ . In classic decentralization models, federal actors put more relative weight on g externalities versus local cost. Sensitivity of policy process to bribery ( $k_1,k_2$ ) surely varies by government. This affects whether or how REDD-concession fees have impact as such payment may filter down from federal to local government levels following particular rules; e.g., across-government transfers from REDD funds I(g) might cover local enforcement costs (e).

Previous modeling suggests lower levels of public goods g, e.g. worse REDD outcomes, are expected if the governments levels do not cooperate closely in setting and enforcing policies, especially when each level has its own binding revenue constraint (B) or separate constituencies. Amacher 2002, for a model like (3), finds that even if government levels coordinate completely, different revenue constraints can lead to greater concession size if revenue needs are important. One idea for addressing such constraints has been to allow government levels to share revenues.

The revenue-constraints issue also leads back to corruption because revenue-constrained central governments may be unable or unwilling to enforce the rules of the concession contracts or agreed revenue sharing or decentralization. There are undoubtedly incentives for corruption that would change the outcomes from REDD payments. Worse, the nature of corruption differs across countries (Karsenty 2008), so best REDD design in one place may not apply to another.

#### 4.2 <u>Reviewing Empirical Evidence</u>

# 4.2.1 Concessions

Often large foreign firms bid for concession contracts stating a volume or area to harvest in a given period and, increasingly, tree selection and logging methods (Karsenty 2008, Cerutti and Tacconi 2008). Illegal logging erodes revenue capture and ability to enforce these practices. Illegal logging within the tropics typically means too much removal outside of the agreed areas and failure to declare the true harvest volume or use agreed methods (Barr 2001; Richards 1999; Hardner and Rice 2000). One common policy idea is to raise royalty rates to raise revenue and to reduce the excess harvest (Vincent 1990, Merry et al. 2002, Palmer 2003). Other ideas include greater enforcement (Pinard and Cropper 2000) as well as area-based, lump-sum royalties (yet Boscolo and Vincent 2007 note that even high area fees could induce unsustainable harvesting).

That better design can improve outcomes is easy to imagine but the exact actions required neither are clear nor are well studied over the full range of conditions within the forested tropics. Greater enforcement, e.g., swims against the historical tides; several studies document low rates of detection and fine collection (Contreras-Hermosilla 2002 for a country-by-country review). In light of the important budget constraint B in (3), REDD payments might not only lower pressure to increase harvests for royalties but also permit more spending on enforcement (Amacher 1999).

In this role, REDD payments would function like debt relief, which Kahn and McDonald 1995 suggest can lower deforestation. As unlike past debt relief (but as I(g) implies in (3)) such REDD payments could be conditioned on forest outcomes through the establishment of baselines as well as remote monitoring, this could further shift local policy incentives. On the other hand, any new protected areas created for REDD are just concessions, i.e. particular designations for the use of public lands with specified outputs (carbon storage versus timber), and corruption issues remain.

# 4.2.2 Corruption

Previous studies have found clear evidence of corruption in forest sectors and associated pressure on the forest from the resulting illegal activities in Africa, in Asia and in Latin America (Contreras-Hermosilla 2002). Bribes have been singled out as a factor that confounds attempts to promulgate improved forest policies; e.g., officials accept bribes for allowing illegal timber trade (Palmer 2005 and Smith et al. 2003 for Asian cases, Siebert and Elwert 2004 for cases in Africa).

A large economics literature has considered when, and how, corruption could be reduced. Considering first when it exists, Jain (2001) sees more corruption when there are higher rents to government-owned resources and low probability of detection and punishment. This implies that corruption can be limited by a well-financed enforcement system. Jain suggests that high wages for public officials can be effective in curbing corruption if there is a relatively high probability of detection. Yet that could raise the debt burden and lower the ability to enforce property rights. Contreras-Hermosilla (2002) finds high corruption with underpaid government forest inspectors, complex regulations around property rights, many bureaucratic steps in obtaining permits to use or establish productive forests, low penalties for illegal logging, and open-access native forests.

Despite some evidence, specific mechanisms by which corruption leads to deforestation need further study, even though it may be widely accepted that 'rule of law' affects deforestation (Deacon 1994). One example of policy-corruption interaction is that shifting royalty function (R) or reducing concession size may be circumvented through adjusted bribes. Analogs could apply to REDD. Better understanding how policy is undermined by corruption surely will be relevant.

# 4.2.3 Decentralization

Most tropical forests are state-owned, so millions who live in them have "use" rights but no legal right to own, manage or block others' exploitation of forest. Recently governments have devolved forest ownership and management to local institutions. As of 2001, at least 60 countries had reported some decentralization reforms in natural resources (Agrawal 2001) and increasingly this includes granting to local communities property rights to the forests (Sunderlin et al. 2008).

Individual or state ownership were long the only forest options considered but there is increasing interest in property rights regimes with management responsibility held by groups of forest users (Ostrom 1990). Common property regimes with local controls in principle can be the most efficient management, given group monitoring and enforcement or higher productivity for large units (McKean 2002). Decentralization could increase the participation and accountability of government (Larson 2004). Also, as noted concerning the  $\varepsilon$  in (1), local actors may have better information on conditions and preferences (Andersson and Lehoucq 2006), though whether local

institutions are accountable to all locals is critical (Ribot 2002; Ribot and Larson 2005) as states unable or unwilling to enforce contracts leave communities vulnerable (Engel and Palmer 2008).

Concerning decentralized forest-management success, in Amazonia indigenous areas are managing to exclude people, even within relatively highly pressured areas (Nepstad et al. 2006). Also, forests under community management in India have been sustainably managed for decades (Agrawal 2001) and there is evidence that community councils conserve forest more effectively than state agencies (Somanathan et al. 2009 and see results from Nepal, Mexico and Nicaragua).

Yet decentralization is not a panacea and clearly it can increase deforestation if attention is not paid to all key incentives. There is no reason to assume that traditional communities all are inherently focused on conservation (Fearnside 2005b, Asner et al. 2005). Rapid and destructive logging can occur, e.g. if communities' tenure or rights are not secure and residents think rights will be revoked. This encourages them to grab forest profits while they can (Oyono 2004, 2005).

Concerning REDD, decentralization and tenure are two-edged swords. A secure tenure that raises willingness to invest in the future could lead either to more forest or to more oil palm. Secure devolution that raises local control could lead to either more forest management or more clearing of the forest. As in each of our models (1), (2) and (3), incentives are key. Consider, for instance, the implications for REDD of the Brazilian Amazon state of Pará being split into three following a recent plebiscite. Is it clear three states would lower forest loss relative to one state?

Bringing this back to concessions and to corruption, i.e. the other foci of this sub-section, REDD success from decentralization is likely to require high levels of cooperation across levels of government. REDD may shift the policy environment but the related past outcomes are mixed. For instance, revenue-sharing agreements between government entities are common for publicly owned forests but they do not always work (Floyd 1999). Smith (2003) argues that in Indonesia the allocation of harvesting permits by district government rarely is consistent with the intentions of the national government. For Africa, Siebert and Elwert (2004) note how central government officials in Benin are resistant to local controls and power over rent capture from the harvesting of government forests, legal and illegal, that would follow from reallocation by decentralization.

# 5. Discussion

# 5.1 Neither Too Easy Nor Too Hard

Among many positions, there are certainly members of two opposite schools of thought regarding a role for policies to reduce forest loss. One school appears to at least implicitly assert that to stop deforestation one need merely pay the opportunity cost of foregone land uses (e.g., profits not being generated by agriculture). This supports a focus on regions where the profit from forest loss is perceived to be low and thus a conclusion that REDD will be relatively cheap.

An opposing school of thought at least implicitly asserts that more is required for REDD, with the implication that the actual total cost is considerably higher (though often not quantified). Underlying this view is pessimism about the possibilities for actually reducing deforestation, often based in part upon the significant failures of many previous policy efforts (see above) and for many countries the lack of governance capacity to effectively implement appropriate policy.

We believe reality lies between these extremes. There are indeed opportunities to avoid tropical deforestation at relatively low cost compared to other options for reducing C emissions. Yet claims about conservation opportunities immediately available have likely been overstated. The costs of reducing deforestation include costs to reform land tenure, to distribute payments, and to establish, manage, and monitor protected areas. Continued demand for commodities due in part to population pressures, weak governance, and other institutional factors constrain options for reducing deforestation. Opportunity costs of land are merely a lower bound on REDD's costs.

That may sound like the second school of thought. Indeed, our review of the past efforts is sobering: many policies did not target, address or even clearly identify critical incentives; interventions lacked local engagement and stakeholder participation; weak governance and lack of title or enforcement limited effectiveness; and the many forest conservation programs almost universally lacked self-evaluation mechanisms, limiting learning and incremental improvement. However, many features of past policies could be drastically improved. Future REDD policies could avoid historical defaults and lead local actors to value additional forest services, making it profitable to manage the forest for the many local and global goods that intact forests provide.

# 5.2 Designing For REDD Feasibility

A prospect of rewards for international forest carbon conservation under future U.S. and international climate policies has brought new energy to the protection of tropical forests. Yet the debate has not been informed by close consideration of all the international and domestic policies required if REDD is to play a large role. We believe that international and domestic interventions can lower deforestation given both the support of local actors and well-informed policy designs.

Past failures suggest gains from designing to encourage locally appropriate interventions. For instance, if comprehensive monitoring can accurately capture GHG emissions reduction, the requirements and incentives can be based on that aggregate outcome and many other details may be left to local actors better placed to significantly and sustainably shift relevant local processes. Generally, consultation with those affected can aid in finding effective and sustainable policies.

# Moving forward:

- the U.S., in concert with international actors, can help forested countries with the costs of conserving forest carbon, including with costs of strengthening the relevant institutions
- international forest carbon policies can adopt performance indicators so that incentives can be effectively applied; monitoring and evaluation will permit ongoing learning
- forested countries can rethink not only forest policy but also how agriculture and infrastructure policies affect forests; strategies will differ as a function of context
- international and domestic actors can re-examine whether actions work well in concert, e.g., policy influences on commodity demands vs. subsidies for agriculture or biofuels

In summary, it is possible to identify incentives driving deforestation and to align local, regional, national, and international incentives in many settings. Climate protection provides a new way for forest protection to contribute, and to succeed, if we learn lessons from the past.

# References

- Agrawal, A. 2001. The regulatory community: Decentralization and the environment in the Van Panchayats (forest councils) of Kumaon, India. Mountain Research and Development, 21(3): 208-211.
- Ahrends, Antje, et al. 2010. Predictable Waves of Sequential Forest Degradation and Biodiversity Loss Spreading from an African City. PNAS, Vol 107, No. 33
- Alix-Garcia, J., Shapiro, E., and Sims, K. 2011a. "Forest Conservation and Slippage: Evidence from Mexico's National Payments for Ecosystem Services Program" Working Paper University of Wisconsin, Madison, mimeo.
- Alix-Garcia, J., C. McIntosh, K.R.E. Sims and J.R. Welch 2011b. "The Ecological Footprint of Poverty Alleviation: evidence from Mexico's Oportunidades Program". University of Wisconsin, Madison, mimeo.
- Alston, L., G. Libecap, and B. Mueller. 2000. Land reform policies, the sources of violent conflict, and implications for deforestation in the Brazilian Amazon. Journal of Environmental Economics and Management 39:162-188.
- Amacher, G., W. Hyde and B. Joshee. 1993. Joint production and consumption in traditional households: fuelwood and crop residues in two districts of Nepal. Journal of Development Studies 30:206–225.
- Amacher, G. 1999. Government preferences and public forest harvesting: A second best approach. American Journal of Agricultural Economics 81(1): 14-28.
- Amacher, G. 2002. Forest Policies and many governments, Forest Science 48:146-158.
- Amacher, G.S., W.F. Hyde, and K.R. Kanel. 1999. Nepali Fuelwood Production and Consumption: Regional and Household Distinctions, Substitution and Successful Intervention. J. Dev. Stud. 35(4):138-163.
- Amacher, G., E. Koskela, and M. Ollikainen. 2007. Royalty reform and illegal reporting of harvest volumes under alternative penalty schemes Environmental and Resource Economics 8(2): 189-211
- Amacher, G., E. Koskela, and M. Ollikainen. 2008. Deforestation and property rights risks. Environment and Development Economics (In Press).
- Amacher, G., M. Ollikainen, and E. Koskela. 2011. Corruption and forest concessions, Journal of Environmental Economics and Management, doi:10:1016/j.jeem.2011.05.007
- Andam, K., P. Ferraro, A. Pfaff, J. Robalino and A. Sanchez (2008). Measuring the effectiveness of protected-area networks in reducing deforestation. PNAS 105(42): 16089–16094.
- Andam, K.S., P. Ferraro, K. Sims, A. Healyd and M. Holland 2010. Protected areas reduced poverty in Costa Rica and Thailand. PNAS 107(22): 9996-10001.
- Andersen, L.E., C.W.J. Granger, E.J. Reis, D. Weinhold, and S. Wunder (2002). The Dynamics of Deforestation and Economic Growth in the Brazilian Amazon, Cambridge University Press.
- Andersson, K.P., and F. Lehoucq. 2006. Municipal Politics and Forest Governance: Comparative Analysis of Decentralization in Bolivia and Guatemala. World Development 34(3): 576–595.

- Angelsen, Arild. 1999. Agricultural Expansion and Deforestation: Modelling the Impact of Population, Market Forces and Property Rights. *Journal of Development Economics* 58:185-218.
- Angelsen, A. and T.K. Rudel 2011. "Implementing REDD+ amidst a forest transition: the policy challenges". Submitted paper for this Symposium.
- Araujo, C., C.A. Bonjean, J. Combes, P. Motel, and E. Reis. 2009. Property rights and deforestation in the Brazilian Amazon. Ecological Economics 68(8-9): 2461-2468.
- Arima, E.Y., R.T. Walker, M. Sales, D. Souza, and S.G. Perz. 2008. The Fragmentation of Space in the Amazon Basin: Emergent Road Networks. Photogrammetric Engineering & Remote Sensing 74 (6): 699–709.
- Armsberg, J. 1998. Economic parameters of deforestation, World Bank Economic Review 12: 133–153.
- Arnold, J., G. Kohlin, and R. Persson. 2006. Woodfuels, livelihoods, and policy interventions: changing perspectives. World Development 34: 596–611.
- Arriagada, R, EO Sills, SK Pattanayak, and PJ Ferraro. 2009. Combining Qualitative and Quantitative Methods to Evaluate Participation in Costa Rica's Program of Payments for Environmental Services. Journal of Sustainable Forestry 28(3): 343-367.
- Arriagada, R., P. Ferraro, E.O.Sills, S.P. Pattanayak, S. Cordero 2011 forthcoming. "Do payments for environmental services affect forest cover? A farm-level evaluation from Costa Rica" Land Economics.
- Asner, Gregory P., David E. Knapp, Eben N. Broadbent, Paulo J.C. Oliveira, Michael Keller, and Jose N. Silva. 2005. Selective Logging in the Brazilian Amazon. Science 310: 480–481.
- Atmadja, S. and E. Sills 2008. Meta-Analysis of Deforestation & Poverty: What is the Consensus? *European Association of Environmental and Resource Economists*, Gothenburg, Sweden, June.
- Ayalew Ali, D., et al. 2011. "Property Rights in a Very Poor Country: Tenure Insecurity and Investment in Ethiopia." Agricultural Economics 42(2011) 75-86.
- Baland, Jean-Marie and Jean-Philippe Platteau, 1996, Halting Degradation of Natural Resources: Is there a Role for Rural Communities?, FAO, Oxford University Press.
- Baland, J. M., Bardhan, P., Das, S., & Mookherjee, D. (2010). The environmental impact of poverty: Evidence from firewood collection in rural Nepal. Economic Development and Cultural Change, 59(1).
- Barbier, E., and J. Burgess. 1996. Economic Analysis of Deforestation in Mexico. Environment and Development Economics 1: 203–239.
- Barbier, E., and J. Burgess. 2001. Tropical deforestation, tenure insecurity and unsustainability, Forest Science 47: 497–509.
- Barbieri A. and D. Carr. 2005. Gender-specific out-migration, deforestation and urbanization in the Ecuadorian Amazon Global and Planetary Change 47: 99–110.
- Barnes, D.F., Krutilla, K., and Hyde, W. (2002). The urban energy transition: energy, poverty and the environment in the developing world. World Bank, Washington, DC (Draft).

- Barr, C. 2001. Timber Concession Reform: Questioning the Sustainable Logging Paradigm. Chapter 4 in Banking on Sustainability: Structural Adjustment and Policy Reform in Post Suharto Indonesia. Center for International Forestry Research (CIFOR), Bogor, Indonesia.
- Barrett C.B. 1999. Stochastic food prices and slash-and-burn agriculture. Environment and Development Economics 4: 161–176.
- Barreto, P. 2008. Quem é dono da Amazônia? Uma análise do recadastramento de imóveis rurais. Belém: Imazon.
- Bauch, S. 2010. Evaluation of conservation policies in the Brazilian Amazon. PhD Dissertation. NC State U., Raleigh NC. *repository.lib.ncsu.edu/ir/bitstream/1840.16/6155/1/etd.pdf*
- Besley T. 1995. Savings, credit and insurance, Chapter 36 in: J. Behrman and T.N. Srinivasan, Editor(s), Handbook of Development Economics, Elsevier, Volume 3, Part A, Pages 2123-2207.
- Bhattamishra, R. and C. B. Barrett. 2010. Community-Based Risk Management Arrangements: A Review. World Development 38 (7): 923-932.
- Bilsborrow, R. 2002: Migration, population change, and the rural environment. Population, Environmental Change, and Security Working Paper 2. Ann Arbor, MI, and Washington, DC: Population and Environment Fellows Program, University of Michigan and Woodrow Wilson International Center.
- Blaser, J., and J. Douglas. 2000. A future for forests? issues and implications for the emerging forest policy and strategy of the World Bank, ITTO Tropical Forest Update 10: 9–14.
- Blom, B., T. Sunderland and D. Murdiyarso. 2010. Getting REDD to work locally: Lessons learned from integrated conservation and development projects. Environmental Science & Policy 13(2): 164-172.
- Bohn, H., and R. Deacon. 2000. Ownership risk, investment, and the use of natural resources, American Economic Review 90: 526–549.
- Boscolo, M., and J.R. Vincent. 2007. Area fees and logging in tropical timber concessions, Environment and Development Economics, 12: 505–520
- Brandon, K. and M. Wells. 2009. Lessons for REDD+ from protected areas and integrated conservation and development projects. In Realising REDD+: National Strategy and Policy Options; Angelsen, A., Ed.; CIFOR: Bogor, Indonesia.
- Brandão Jr., A.O., and C.M. Souza Jr. 2006. Mapping unofficial roads with Landsat images: A new tool to improve the monitoring of the Brazilian Amazon rainforest. International Journal of Remote Sensing 27: 177–189.
- Bray, D.B., Merino-Pe' rez, L., Negreros-Castillo, P., Segura-Warnholtz, G., Torres-Rojo, J.M., Vester, H.F.M., 2003. Mexico's community-managed forests as a global model for sustainable landscapes. Conservation Biology 17, 672–677.
- Brito, B. and P. Barreto. 2009. The risks and the principles for landholding regularization in the Amazon. State of the Amazon N° 10. Imazon, Belém, Brazil.
- Brooks JS, Franzen MA, Holmes CM, Grote MN, Borgerhoff Mulder M. 2006. Testing hypotheses for the success of different conservation strategies. Conserv. Biol. 20(5):1528-38

- Brown DR (2008) A spatiotemporal model of shifting cultivation and forest cover dynamics. Environ Dev Econ 13:643–671.
- Caldas, M., Walker, R., Arima, E., Perz, S., Aldrich, S., & Simmons, C. (2007). Theorizing Land Cover and Land Use Change: The Peasant Economy of Amazonian Deforestation. *Annals of the Association of American Geographers*, 97(1), 86-110. doi:10.1111/j.1467-8306.2007.00525.x
- Campbell, B.M., Sayer, J.A. & Walker, B. 2010. Navigating trade-offs: Working for conservation and development outcomes. Ecology and Society 15(2), available from http://www.ecologyandsociety.org/vol15/iss2/art16/
- Cardenas, Juan Camilo, John K. Stranlund, and Cleve E. Willis. 2000. Local Environmental Control and Institutional Crowding-out. World Development 28(10): 1719-1733.
- Carr, D. 2004. Proximate Population Factors and Deforestation in Tropical Agricultural Frontiers Population and Environment, 25 (6): 585-612.
- Carr, D., W. Pan, and R. Bilsborrow. 2006. Declining fertility on the frontier: the Ecuadorian Amazon. Population and Environment. 28(1): 17–39. doi: 10.1007/s11111-007-0032-y.
- Carter, M.R., and Y. Yao. 2002. Local versus Global Separability in Agricultural Household Models: The Factor Price Equalization Effect of Land Transfer Rights. Am. J. Agr. Econ. 84(3):702-715.
- Caviglia, Jill and E. Sills. 2005. Livelihood Strategies of Traditional and Migrant Populations in the Brazilian Amazon: Diversification and Deforestation. *Agricultural Economics*. 32: 221–237.
- Caviglia-Harris, Jill L., Sills, Erin O., Jones, Luke, Saha, Shubhayu, Harris, Daniel, McArdle, Suzanne, Roberts, Dar, Pedlowski, Marcos and Powell, Rebecca. 2009. Modeling land use and land cover change in an Amazonian frontier settlement: strategies for addressing population change and panel attrition. Journal of Land Use Science, 4: 4, 275 307.
- Cerutti, P. O., R. Nasi, and L. Tacconi. 2008. Sustainable forest management in Cameroon needs more than approved forest management plans. Ecology and Society 13(2): 36. [online] URL: http://www.ecologyandsociety.org/vol13/iss2/art36/
- Cerutti, P.O. and L. Tacconi 2008. Forests, Illegality, and Livelihoods: The Case of Cameroon. Society & Natural Resources 21 (9) October: 845 – 853.
- Chomitz, K., and D. Gray 1996. Roads, land use, and deforestation: a spatial model applied to Belize, World Bank Economic Review 10: 487–512.
- Clarke, H., W. Reed, and R. Shrestra. 1993. Optimal enforcement of property rights on developing country forests sucject to illegal logging. Resource and Energy Economics 15:271-293.
- Conde, D., and A. Pfaff (2008). Sequenced Road Investments and Clearing of the Mayan Forest. Duke University mimeo.
- Contreras-Hermosilla, A. 2000. The underlying causes of forest decline, CIFOR Occasional Paper #30, June, 29pp.

- Contreras-Hermosilla, A. 2002. Law Compliance in the Forestry Sector: An Overview, World Bank Institute, 2002.
- Cooke, P. A. 1998. Intrahousehold Labor Allocation Responses to Environmental Good Scarcity. Econ. Dev. Cult. Change 46(4) :807-830.
- Cooke, P., G. Kohlin, and W. Hyde. 2008. Fuelwood, forests, and community management evidence from household studies. Environment and Development Economics 13: 103–135.
- Coomes, O. T., B. L. Barham, and Y. Takasaki. 2004. Targeting conservation–development initiatives in tropical forests: Insights from analyses of rain forest use and economic reliance among Amazonian peasants. Ecological Economics 51:47–64.
- Coxhead, Ian, Agnes Rola, and Kwansoo Kim. 2001. How do National Markets and Price Policies Affect Land Use at the Forest Margin? Evidence from the Philippines. *Land Economics* 77 (2):250-267.
- de Sherbinin, A., Vanwey, L., McSweeney, K., Aggarwal, R., Barbieri, A., Henry, S., Hunter, L. M., et al. (2008). Rural Household Demographics, Livelihoods and the Environment. *Global environmental change : human and policy dimensions*, 18(1), 38-53. doi:10.1016/j.gloenvcha.2007.05.005
- Deacon, Robert T. (1994). "Deforestation and the Rule of Law in a Cross-Section of Countries". Land Economics 70(4): 414-430.
- Deininger, Klaus, and Songqing Jin. 2006. Tenure Security and Land-related investment: Evidence from Ethiopia. European Economic Review, 50 (2006) 1245-1277.
- Delgado, C., A. Pfaff et al. (2008). "Will Nearby Protected Areas Constrain Road Impacts On Deforestation?". Presentation at NASA LBA conference Amazon In Perspective, Manaus.
- Delgado, C., D.A. Conde, J.O. Sexton, F. Colchero, J.J. Swensen, and A. Pfaff (2010). Deforestation dynamics in response to the evolution of the Western Amazonian Inter-Oceanic Highway. Duke University (submitted).
- Ehui, S.K. and T.W. Hertel 1989. "Deforestation and Agricultural Productivity in the Cote d'Ivoire". AJAE 71(3):703-11.
- Engel, S., and C. Palmer. 2008. Payments for environmental services as an alternative to logging under weak property rights: The case of Indonesia. Ecological Economics (65): 799–809.
- Fearnside, Philip M. 2005a. Conservation Policy in Brazilian Amazonia: Understanding the Dilemmas. World Development 31(5): 757–779.
- Fearnside, P.M. 2005b. Deforestation in Brazilian Amazonia: History, Rates, and Consequences. Conserv Biol 19:680–688
- Fearnside, P.M. 2005c. Indigenous peoples as providers of environmental services in Amazonia: Warning signs from Mato Grosso. pp. 187–198. In: A. Hall (ed.) Global Impact, Local Action: New Environmental Policy in Latin America, University of London, School of Advanced Studies, Institute for the Study of the Americas, London, U.K. 321 pp.
- Ferraro, P.J., R.D. Simpson. 2002. The Cost-Effectiveness of Conservation Payments. Land Economics. 78: 339-353.

- Finer M, Jenkins CN, Pimm SL, Keane B, Ross C (2008) Oil and Gas Projects in the Western Amazon: Threats to Wilderness, Biodiversity, and Indigenous Peoples. PLoS ONE 3(8): e2932. doi: 10.1371/journal.pone.0002932
- Fisher, M.; Shively, G. E.; Buccalo, S. 2005. Activity choice, labor allocation, and forest use in Malawi, Land Econ 81(4): 503-517.
- Fisher M, M Chaudhury, B McCusker. 2010. Do Forests Help Rural Households Adapt to Climate Variability? Evidence from Southern Malawi. World Development doi:10.1016/j.worlddev.2010.03.005
- Floyd, D. 1999, Forests of discord: options for governing our national forests and federal public lands, Society of American Foresters.
- Foster, A.D. and M.D. Rosenzweig. 2003. Economic Growth and the Rise of Forests. The Quarterly Journal of Economics 118: 601-637.
- Fox, Jefferson, R.R. Rindfuss, S.J. Walsh, and V. Mishra (eds.) 2003. People and the Environment: Approaches for Linking Household and Community Surveys to Remote Sensing and GIS. Dordrecht: Kluwer Academic Publishing Group.
- Fu Y, J Chen, H Guo, A Chen, J Cui, H Hu. 2009. The role of non-timber forest products during agroecosystem shift in Xishuangbanna, southwestern China. Forest Policy and Economics 11(1): 18-25.
- Garnett, S.T., J. Sayer, J. Du Toit 2007. "Improving the Effectiveness of Interventions to Balance Conservation and Development: a Conceptual Framework." Ecology and Society 12: 2.
- Geist, H.J., and E.F. Lambin. 2002. Proximate Causes and Underlying Driving Forces of Tropical Deforestation. BioScience 52(2): 143–150.
- Gray, J. 2002. Forest Concession Policies and Revenue Systems: Country Experience and Policy Changes for Sustainable Tropical Forestry. World Bank Technical Paper No. 522, Forest Series, Washington D.C.
- Guertin, C. 2003. Illegal Logging and Illegal Activities in the Forest Sector: Overview and Possible Issues for the UNECE Timber Committee and FAO European Forestry Commission. Presented at UNECE Timber Committee Market Discussions, October 7–8, 2003, Geneva.
- Hardner, J., and R. Rice. 2000. Rethinking Forest Use Contracts in Latin America. Environment Division WP. Interamerican Development Bank, Washington D.C.
- Heltberg, R., T. Arndt, and N. Sekhar. 2000. "Fuelwood consumption and forest degradation: a household model for domestic energy substitution in Rural India", Land Econ 76: 213–232
- Honey-Rosés, J. 2009. Illegal Logging in Common Property Forests. Society & Natural Resources 22(10): 916 930.
- Hyde, W. F. and G. Köhlin. 2000. "Social Forestry reconsidered", Silva Fennica 34(3):285-314.
- Jagger, P., E. Sills, K. Lawlor, and W. Sunderlin. 2010. A guide to Learning about the Livelihood Impacts of REDD+. CIFOR Working Paper #56, available from: http://www.cifor.cgiar.org/nc/online-library/browse/view-publication/publication/3283.html
- Jain A.K. (2001). 'Corruption: a review', Journal of Economic Surveys, vol. 15(1), pp. 71–121.

- Joppa, L., and A. Pfaff. 2009. High & Far: Biases in the location of protected areas. PLoS ONE 4(12): e8273. doi: 10.1371/journal.pone.0008273
- Joppa, L., and A. Pfaff. 2010a. Re-assessing the forest impacts of protection: the challenge of non-random protection & a corrective method. Annals of the New York Academy of Sciences 1185: 135–149.
- Joppa, L., and A. Pfaff. 2010b. Global Park Impacts: how could protected areas avoid more deforestation? Proceedings of the Royal Society B, doi:10.1098/rspb.2010.1713
- Kahn, J., and J. McDonald. 1995. Third-world debt and tropical deforestation, Ecological Economics 12: 107–123.
- Kaimowitz, D., and A. Angelsen. 1998. Economic Models of Tropical Deforestation: A Review. Center for International Forestry Research. Jakarta.
- Karsenty, A. 2008. The architecture of proposed REDD schemes after Bali: Facing critical choices. International Forestry Review 10: 443–457.
- Keohane R.O and Levy, M.A. (Eds.), 1996. Institutions for Environmental Aid, Global Environmental Accord Series; MIT, Boston, MA.
- Kerr, S. and C. Streck 2011. "The economics of international policies to reduce emissions from deforestation and degradation". Submitted paper in this Symposium.
- Klasen, S., Faust, H., Grimm, M., & Schwarze, S. (2010). Tropical Rainforests and Agroforests under Global Change. (T. Tscharntke, C. Leuschner, E. Veldkamp, H. Faust, E. Guhardja, & A. Bidin, Eds.) Science, 213-236. Berlin, Heidelberg: Springer Berlin Heidelberg.
- Köhlin, G. (1998), 'The value of social forestry in Orissa, India', Unpublished Ph.D. thesis, Department of Economics, Goteborg University.
- Kohlin, G., and G. Amacher. 2005. Welfare implications of social forestry projects: the case of Orissa India. American Journal of Agricultural Economics 87(4): 855–869.
- Köhlin, G., E. Sills, S. Pattanayak, and C. Wilfong. 2011. Energy, Gender and Development What are the Linkages? Where is the Evidence? World Bank Policy Research Working Paper 5800.
- Krüger, O. 2005. The role of ecotourism in conservation: panacea or Pandora's box? Biodiversity and Conservation 14: 579–600. DOI 10.1007/s10531-004-3917-4
- Laporte N. T., J.A. Stabach, R. Grosch, T.S. Lin and S.J. Goetz. 2007. Expansion of Industrial Logging in Central Africa Science 316 (5830): 1451.
- Larson, A.M. 2004. Democratic Decentralisation in the Forestry Sector: Lessons Learned from Africa, Asia and Latin America.
- Leisher C., M. Sanjayan, J. Blockhus, A. Kontoleon, and S. Larsen. 2010. Does Conserving Biodiversity Work to Reduce Poverty? A State of Knowledge Review. August.
- Linde-Rahr, M. 2003. "Property rights and deforestation: the choice of fuelwood source in rural Vietnam", Land Economics 79: 217–234.

- Lopez, R., and G. Galinato. 2005. Deforestation and Forest-Induced Carbon Dioxide Emissions in Tropical Countries: How Do Governance and Trade Openness Affect the Forest-Income Relationship? Journal of Environment and Development 14(1): 73–100.
- Lubowski. R and S. Rose 2011. Submitted paper in this Symposium.
- Maertens, M., Zeller, M., & Birner, R. (2006). Sustainable agricultural intensification in forest frontier areas. *Agricultural Economics*, *34*(2), 197-206. doi:10.1111/j.1574-0864.2006.00118.x
- Marshall E, Schreckenberg, Newton AC. 2006. Commercialization of non-timber forest products: Factors influencing success. UNEP, World Conservation Monitoring Center, Cambridge, UK.
- McKean, M.A. 2002. Common Property and Coaseian Bargains. Unpublished manuscript.
- McShane. T.O., and M. P. Wells. 2004. "Integrated conservation and development?" In Getting biodiversity projects to work: towards better conservation and development, ed. T.O. McShane and M.P. Wells. Columbia University Press, New York: Columbia University Press.
- McSweeney K. 2005. Natural Insurance, Forest Access, and Compound Misfortune: Forest Resources in Smallholder Coping Strategies Before and After Hurricane Mitch, Northeastern Honduras. World Development, 33(9): 1453-1471
- Merry, F., and G. Amacher. 2005. Forest concessions and policy choices for the Brazilian Amazon. Journal of Sustainable Forestry 20(2): 15–44.
- Merry, F., G. Amacher, B. Pokorny, E. Lima, E. Scholz, and D. Nepstad. 2002. Some doubts about forest concessions in the Brazilian Amazon. 2003. ITTO Tropical Forestry Update
- Merry, F. D., Sheikh, P. a, & Mcgrath, D. G. (2004). The role of informal contracts in the growth of small cattle herds on the floodplains of the Lower Amazon. *Agriculture and Human Values*, *21*(4), 377-386. doi:10.1007/s10460-003-1246-y
- Merry, F., G. Amacher, D. Nepstad, and E. Lima. 2008. Land values in frontier settlements of the Brazilian Amazon. World Development (In Press).
- Mertens, B., R. Poccard-Chapuis, M.-G. Piketty, A.-E. Laques, A. Venturieri. 2002. Crossing spatial analyses and livestock economics to understand deforestation processes in the Brazilian Amazon: the case of Sao Felix do Xingu in South Para', Agricultural Economics 27: 269–294.
- Meyfroidt, P., T.K. Rudel and E.F. Lambin (2010). Forest transitions, trade, and the global displacement of land use
- Miranda, M., I. Porras, and M. Moreno. 2003. "The social impacts of payments for environmental services in Costa Rica. A quantitative field survey and analysis of the Virilla watershed."London: IIED (processed).
- Morton, D.C., R.S. Defries, J.T. Randerson, L. Giglio, W. Schroeder, and G.R. Van Der Werf. 2008. Agricultural Intensification Increases Deforestation Fire Activity in Amazonia. Global Change Biology 14: 2262–75.

- Munoz, C., A. Guevara, J.M. Torres and J. Brana 2008. "Paying for the hydrological services of Mexico's forests: analysis, negotiations and results. Ecological Economics 65:725-736.
- Muller, J. and H.J. Albers (2004). Enforcement, payments, and development projects near protected areas: how the market setting determines what works where. *Resource and Energy Economics*, *26*(2), 185-204. doi:10.1016/j.reseneeco.2003.11.007
- Nelson, G.C. and D. Hellerstein (1997). "Do roads cause deforestation? using satellite images in econometric analysis of land use". *American Journal of Agricultural Economics* 79:80.
- Naughton-Treves, L., M.B. Holland, K. Brandon. 2005. "The Role of Protected Areas in Conserving Biodiversity and Sustaining Local Livelihoods." Annual Review of Environment and Resources 30: 219-252.
- Nepstad, D., S. Schwartzman, B. Bamberger, M. Santilli, D. Ray, P. Schlesinger, P. Lefebvre, A. Alencar, E. Prinz, G. Fiske, A. Rolla (2006). "Inhibition of Amazon Deforestation and Fire by Parks and Indigenous Lands". Conservation Biology 20(1): p.65–73.
- Ortiz, E., L. Sage, and C. Borge. 2003. "Impacto del programa de pago de servicios ambientales en Costa Rica como medio de reducción de la pobreza en los medios rurales." Serie de Publicaciones RUTA. San José: Unidad Regional de Asistencia Técnica.
- Ostrom, Elinor. 1990. Governing the Commons: The Evolution of Institutions for Collective Action. New York: Cambridge University Press.
- Oyono, R. 2004. One step forward, two steps back ? Paradoxes of natural resources management decentralisation in Cameroon. Journal of Modern African Studies 42(1): 91–111.
- Oyono, P.R. 2005. Profiling Local-Level Outcomes of Environmental Decentralizations: The Case of Cameroon's Forests in the Congo Basin. The Journal of Environment & Development 14(3): p. 317–337.
- Palmer, C. 2003. The Extent and Causes of Illegal Logging: an Analysis of a Major Cause of Tropical Deforestation in Indonesia. Center for Social and Economics Research, University College London, ISSN 0967-8875.
- Palmer, C.E. 2005. "The Nature of Corruption in Forest Management". World Econ 6(2), 1-10.
- Pan, W., D. Carr, A. Barbieri, R. Bilsborrow, and C. Suchindran. 2007. Forest clearing in the Equadorian Amazon: a study of patterns over space and time. Population Research and Policy Review 26: 635–659.
- Pattanayak, S. K., and E. Sills. 2001. "Do Tropical Forests Provide Natural Insurance? Non-Timber Forest Product Collection in the Brazilian Amazon," Land Economics 77(4): 595-612.
- Pattanayak, Subhrendu, E. Sills, and R. Kramer. 2004. Seeing the Forests for the Fuel. Environment and Development Economics. 9(2): 155-179.
- Pattanayak, S.K., S. Wunder and P.J. Ferraro 2010. "Show Me The Money: do payments supply environmental services in developing countries?". REEP 2010:1-21 doi:10.1093/reep/req006
- Paumgarten F. 2005. The role of non-timber forest products as safety-nets: A review of evidence with a focus on South Africa. Geojournal, 64: 189-197.
- Pendleton, L. H., & Howe, E. L. (2002). Market Integration, Development, and Smallholder Forest Clearance. *Land Economics*, 78(1), 1. doi:10.2307/3146919

- Perz, S. G.; Aramburú, C.; Bremner J. 2005. Population, land use and deforestation in the Pan Amazon basin: a comparison of Brazil, Bolivia, Colombia, Ecuador, Perú and Venezuela. Environment, Development and Sustainability, v. 7: 23- 49.
- Perz, S.G., C. Overdevest, E.Y. Arima, M.M. Caldas, and R.T. Walker. 2007. Unofficial road building in the Brazilian Amazon: dilemmas and models of road governance. Environmental Conservation 34: (2):112–121 Cambridge University Press
- Pfaff A. 1999. What Drives Deforestation in the Brazilian Amazon? Evidence from Satellite and Socioeconomic Data, Journal of Environmental Economics and Management 37(1): 26–43.
- Pfaff A., J.A. Robalino, R. Walker, E. Reis, S. Perz, C. Bohrer, S. Aldrich, E. Arima, M. Caldas, W Laurance and K. Kirby. 2007. Road Investments, Spatial Intensification and Deforestation in the Brazilian Amazon. Journal of Regional Science 47: 109–123.
- Pfaff, A., J.A. Robalino, G.A. Sánchez-Azofeifa, K. Andam and P. Ferraro 2009. Park Location Affects Forest Protection: Land Characteristics Cause Differences in Park Impacts across Costa Rica, The B.E. Journal of Economic Analysis & Policy: Vol. 9: Iss. 2 (Contributions), Article 5 (available at: http://www.bepress.com/bejeap/vol9/iss2/art5).
- Pfaff A. and R. Walker 2010. "Regional Interdependence and Forest 'Transitions': substitute deforestation limits the relevance of local reversals". Land Use Policy 27:119-129.
- Pfaff A., R. Walker, S. Perz, W. Laurance, C. Bohrer, J.A. Robalino, S. Aldrich, E. Arima, M. Caldas and K. Kirby 2011a (revise/resubmit). Heterogeneous Transport Impacts and 'Clean' Development. Mimeo, Duke University.
- Pfaff, A., J. Robalino and D. Herrera 2011b. Roads' Heterogeneous Forest Impacts: Brazilian Amazon frontier spatial dynamics. ASSA/AERE sesssion presentation, January, Denver.
- Pfaff, A., J.A. Robalino and L.D. Herrera et al. 2011c. "Might REDDer Be Better? Protected Areas' Highly Varied Forest Impacts". Mimeo, Duke University. Presentation at WCERE.
- Pfaff, A., J.A. Robalino, L.D. Herrera, E. Lima and C. Sandoval 2011d. "Estimating Avoided Deforestation by Protection Type: can governance plus location equal forest plus livelihoods?". Mimeo, Duke University. Presentation at UW LTC conference, Madison.
- Pfaff, A., J.A. Robalino and L.D. Herrera et al. 2011e. "Decentralization Given Environment-Development Tradeoffs: federal versus state conservation and impacts on Amazonian deforestation". Mimeo, Duke University. Presentation at AERE conference, Seattle.
- Pfaff, A., J. Robalino and D. Herrera et al. 2011f. "Protected Areas' Forest Spillovers in the Brazilian Amazon: what might explain 'blockage'?". Mimeo, Duke University.
- Pinard, M. A. and Cropper, W. P.: 2000, 'Simulated effects of logging on carbon storage in dipterocarp forest', J. Appl. Ecol. 37, 267.
- Putz. F.E., D.O. Dykstra, and R. Heinrich. 2000. Why Poor Logging Practices Persist in the Tropics. Conservation Biology 14(4): 951-956.
- Ribot, Jesse C., and Anne M. Larson. 2005. Democratic Decentralisation Through a Natural Resource Lens. New York: Routledge.

- Ribot, Jesse. 2002. Democratic Decentralization of Natural Resources: Institutionalizing Popular Participation. Washington, DC: World Resources Institute.
- Richards, R. 1999. Internalizing the Externalities of Tropical Forestry: Review of Financing and Incentive Mechanisms. Overseas Development Institute, London.
- Robalino, J. 2007 "Land Conservation Policies and Income Distribution: Who Bears the Burden of our Environmental Efforts?" Environment and Development Economics 12 (4), 521-533
- Robalino, J., A. Pfaff, A. Sanchez, F. Alpizar, C. Leon. 2008. Deforestation Impacts of Environmental Services Payments: Costa Rica's PSA Program 2000–2005. CATIE.
- Robalino, J. and A. Pfaff (2011). Eco-payments and Forest Loss in Costa Rica: all dressed up but nowhere to avoid deforestation. Duke University mimeo (revise/resubmit)
- Robalino, J.A., A. Pfaff, A. Sanchez, L. Villalobos. 2011. "Evaluating Spillover Effects of Land Conservation Policies in Costa Rica". Mimeo, CATIE.
- Robinson, E.J.Z., J.C. Williams, and H.J. Albers. 2002. "The Influence of Markets and Policy on Spatial Patterns of Non-timber Forest Product Extraction," *Land Economics* 78(2), 260–71.
- Robison, E. and R. Lokina. 2011. A spatial-temporal analysis of the impact of access restrictions on forest landscapes and household welfare in Tanzania. Forest Policy and Economics 13(1): 79-85.
- Ross, Michael. 1996. "Conditionality and Logging Reform in the Tropics." In Institutions for
- Rudel, T.K., K. Flesher, D. Bates, S. Baptista, and P. Holmgren. 2000. Tropical deforestation literature: geographical and historical patterns. Unasylva 203 (51): 11–18.
- Sadoulet, E., and A. de Janvry. 1995. Quantitative Development Policy Analysis. Johns Hopkins University Press, Baltimore. 397 p.
- Sanchez-Azofeifa, G., A. Pfaff, J. Robalino and J. Boomhower (2007). "Costa Rica's Payment for Environmental Services Program: Intention, Implementation and Impact". Conservation Biology 21(5):1165-1173.
- Shackleton SE, Shackleton CM. 2004. The importance of non-timber forest products in rural livelihood security and as safety nets: a review of evidence from South Africa. South African Journal of Science 100.
- Shanley P, Luz L, Swingland I (2002) The Faint Promise of a Distant Market: a survey of Belem's trade in non-timber forest products. Biodiversity and Conservation 11:615-636.
- Shively, Gerald. 1998. "Economic Policies and the Environment: The Case of Tree Planting on Low-income Farms in the Philippines." Environment and Development Economics, 3(1998): 83-104.
- Shively, G. E. 2001. "Agricultural Change, Rural Labor Markets, and Forest Clearing: An Illustrative Case from the Philippines." *Land Economics* 77(2): 268-284.
- Siebert, U., and G. Elwert. 2004. Combating corruption and illegal logging in Benin West Africa. Journal of Sustainable Forestry 19:239-261.

- Sierra, R, and E. Russman 2006.On the efficiency of the environmental service payments: A forest conservation assessment in the Osa Peninsula, Costa Rica Ecological Economics 59: 131–141
- Siegmund-Schultze, M., Rischkowsky, B., da Veiga, J. B., & King, J. M. (2007). Cattle are cash generating assets for mixed smallholder farms in the Eastern Amazon. *Agricultural Systems*, *94*(3), 738-749. doi:10.1016/j.agsy.2007.03.005
- Sills, Erin, S. Lele, T. Holmes, and S. Pattanayak. 2003. Role of Nontimber Forest Products in the Rural Household Economy P. 259 281 in E. Sills and K. Abt, eds., Forests in a Market Economy, Kluwer Academic Publishers.
- Sills, E., S. Pattanayak, P. Ferraro, and K. Alger. 2006. Abordagens analíticas na avaliação de impactos reais de programas de conservação. Megadiversidade 2(1-2): 39-49.
- Sills, E. and J. Caviglia-Harris. 2008. Evolution of the Amazonian Frontier: Land Values in Rondônia, Brazil. Land Use Policy. 26: 55–67.
- Sills, E., F. Parmgarten, P. Shanley, and J. de Beer. 2011. Evolving perspectives on non-timber forest products. Chapter 2 in: S. Shackleton, C. Shackleton, and P. Shanley (eds.) Non-timber Forest Products in the Global Context. Springer.
- Singh, I., L. Squire, and J. Strauss (eds). 1986. Agricultural Household Models. Johns Hopkins University Press, Baltimore. 335 p.
- Simmons, C., Walker, R., Perz, S., Aldrich, S., Caldas, M., Pereira, R., Leite, F., Fernandes, L.C., and Arima, E. 2009. Doing it for Themselves: Direct Action Land Reform in the Brazilian Amazon. World Development (2009), doi:10.1016/j.worlddev.2009.06.003
- Smith, J., and K. Obidzinski, and Subarudi and I. Suramenggala. 2003. Illegal Logging, Collusive Corruption and Fragmented Governments in Kalimantan, Indonesia, International Forestry Review, 5(3), 293–302.
- Somanathan, E., R. Prabhakar, B.S. Mehta. 2009. Decentralization for cost-effective conservation. PNAS 106 (11): 4143–4147.
- Sonoda T. and Y. Maruyama. 1999. Effects of the Internal Wage on Output Supply: A Structural Estimation for Japanese Rice Farmers *American Journal of Agricultural Economics* 81 (1): 131-143.
- Sunderlin, W.D., J. Hatcher, M. Liddle. 2008. From Exclusion to Ownership? Challenges and Opportunities in Advancing Forest Tenure Reform. Washington, D.C.: Rights and Resources Initiative.
- Tachibana, T., Nguyen, T.M., Otsuka, K., 2001. Agricultural intensification versus extensification: a case study of deforestation in the northern-hill region of Vietnam. Journal of Environmental Economics and Management 41 (1), 44–69.
- Takasaki Y. 2007. Dynamic household models of forest clearing under distinct land and labor market institutions: can agricultural policies reduce tropical deforestation? Environment and Development Economics 12: 423-443.

- van Kooten, G.C., R.A. Sedjo, and E.H. Bulte. 1999. Tropical deforestation: issues and policies, in Folmer, H., and T. Tietenberg (ed): The International Year Book of Environmental and Resource Economics 1999/2000: 199–248.
- van Soest, D., E. Bulte, A. Angelsen, and G. van Kooten. 2002. Technological change and tropical deforestation: a perspective at the household level. Environment and Development Economics 7: 269–280 DOI:10.1017/S1355770X02000177
- Van't Veld, Klass et al. 2006. "India's Firewood Crisis Re-examined." Resources for the Future, Discussion Paper 06-25
- VanWey, L. K., D'Antona, Á. O., & Brondízio, E. S. (2007). Household demographic change and land use/land cover change in the Brazilian Amazon. *Population and Environment*, 28(3), 163-185. doi:10.1007/s11111-007-0040-y
- Verburg, P.H., K.P. Overmars, and N. Witte. 2004. Accessibility and land-use patterns at the forest fringe in the northeastern part of the Philippines The Geographical Journal 170 (3): 238–255
- Vincent, J. 1990. Rent Capture and the Feasibility of Tropical Forest Management. Land Economics 66: 212–23.
- Walker, R., Perz, S.G., Caldas, M., Teixeira Silva, L.G., 2002. Land use and land cover change in forest frontiers: the role of household life cycles. International Regional Science Review 25 (2), 169–199.
- Weber, J., E. Sills, S. Bauch, and S. Pattanayak. 2011. Do ICDPs work? An empirical evaluation of forest-based microenterprises in the Brazilian Amazon. Land Economics 87 (4): 645–681
- Wibowo, D.H., and R.N. Byron. 1999. Deforestation mechanisms: A survey. International Journal of Social Economics 26(1-2-3): 455–474.
- Wunder S. 2001. Poverty Alleviation and Tropical Forests What Scope for Synergies? World Dev. 29(11):1817-1833
- Zbinden, S., and D.R. Lee. 2005. "Paying for environmental services: An analysis of participation in Costa Rica's PSA Program." World Development, 33(2), pp.255-272.
- Zwane A. P., 2007, "Does poverty constrain deforestation? Econometric evidence from Peru", Journal of Development Economics, 84(1): 330-349.

# **ENDNOTES / APPENDIX**

<sup>i</sup> Payments could affect migration decisions. In a region well integrated into markets, these effects of household consumption would be spatially dissipated. But on the other hand, an increase in income and resulting increase in returns to labor could attract more households to the region. Similar concerns have been raised regarding ICDPs.

<sup>ii</sup> These parameters include household demographics that affect both demand for consumption and supply of labor. Note that in our generic model, we assume that labor is hired only for clearing forest. This is a logical use of hired (non-family) labor, as deforestation effort is easier to monitor than other agricultural or forest production activities. However, the model can also be generalized further to allow for hired labor in all household production activities.

<sup>iii</sup> With multiple periods,  $\Psi_f$  can reflect prior collection that degrades forest and/or builds forest skill and knowledge (Pattanayak and Sills 2001). It could include total forest area available, perhaps a function of the initial land minus A (e.g., for agricultural colonization projects in Brazil) or, for common property or open access, a function of clearing by the entire village. Where households live on the margin of large open access forests, the distance to collect could increase with prior agricultural clearing, but most relevant literature considers competition for labor in agriculture and forest production without the competition for land, essentially assuming zero local opportunity costs of forest.

<sup>iv</sup> Others model two seasons (Zwane 2007), two periods (Takasaki 2007) or multiple periods (Tachibana et al. 2001). This introduces soil depletion and discounting. Because credit markets are typically thin or missing, they are often omitted from such models. Assets  $\Psi$  grow through investment or purchase or may be specified as a technology for intensification. Inputs from forest (e.g. poles, fodder) could enter agricultural production functions (Sills et al. 2003). Land can be differentiated by distance from a residence and primary/secondary forest (Pendleton and Howe 2002) or upland/lowland (Shively 2001, Coxhead et al. 2001, Klasen et al. 2010, Maertens et al. 2006, Tachibana et al. 2001).

<sup>v</sup> External transfers for refraining from deforestation could also be probabilistically conditioned on time allocation, e.g. a chance of not receiving payment if a caught spending time on deforestation (see Muller and Albers 2004).

<sup>vi</sup> On the forest margin, households are likely to have limited off-farm employment opportunities, even if generally well integrated into other markets. This could be represented in such a model by assuming either of the following constraints: (i) households use only their own labor for everything ( $t_w$ =0); (ii) there is a constraint on the amount of hired labor ( $t_w < z$  (Weber et al. 2011)). Various models alternatively assume that  $t_w$  and  $t_D$  are fully substitutable or that hired labor has lower productivity, e.g. due to significant supervisory costs in many tasks. Finally, we have omitted from this basic household budget the possibility of selling cleared land at end of the first period or in future periods. That would create an additional value generated by clearing forest (Takasaki 2007, Caldas et al. 2007).

<sup>vii</sup> In evaluating policy impacts in light of this point, we must also consider the impact of all such characteristics upon the baseline household behavior without policy interventions. For instance, Pendleton and Howe 2002 find that more patient households clear more forest as an investment in future production. Caldas et al. 2007 reviews work at the household level in Brazil, in Peru and in Ecuador which links households' attributes to land use and land cover.

<sup>viii</sup> Community forest management for timber is typically not part of ICDPs, which tend to be focused in and around the protected areas, but instead is more aligned with various initiatives to decentralize forest management. That includes Mexico's ejido system (see Bray et al. 2003, e.g.) as well as India's panchayat system (see Baland and Platteau 1996 as well as Somanathan et al. 2009). <sup>ix</sup> Note that illegal practice surely could involve not only under-reporting harvest level but also misreporting method. Note also that 'leakage' can include logging outside the agreed boundary, which here we assume is not reported in x. Also even if purely reputable firms follow concession terms, 'leakage' can involve other firms following (see 2.2.1).